

The Debt-Contracting Value of Accounting Information and Loan Syndicate Structure

RYAN BALL,* ROBERT M. BUSHMAN,* AND FLORIN P. VASVARI†

Received 8 January 2007; accepted 2 January 2008

ABSTRACT

We investigate how both the ownership structure and explicit contractual structure of syndicated loan deals are shaped by the debt-contracting value (DCV) of borrowers' accounting information. DCV captures the inherent ability of firms' accounting numbers to capture credit quality deterioration in a timely fashion. We hypothesize and document that when a borrower's accounting information possesses higher DCV, information asymmetry between the lead arranger and other syndicate participants is lower, allowing lead arrangers to hold a smaller proportion of new loan deals. Further, we document that the influence of DCV on the proportion of the loan retained is conditional on the lead arranger's reputation, the existence of a credit rating, and the lead arranger's previous relationships with the same borrower. Finally, we find that when loans include performance pricing provisions, the likelihood that the single performance measure used is an accounting ratio, rather than a credit rating, is increasing in DCV.

^{*}Kenan-Flagler Business School, University of North Carolina-Chapel Hill; †London Business School. We thank Sudipta Basu, Bob Kaplan, Krishna Palepu, Tim Ritchie (Head of Global Loans at Barclays Capital), an anonymous referee, and seminar participants at the 2007 *Journal* of Accounting Research Conference, the Duke/UNC Fall Camp, the 2007 Financial Economics and Accounting Conference, Harvard Business School, London Business School, Northwestern University, Rotterdam School of Management, UCLA, University of Southern California, and Wharton School for helpful comments. We also thank Moody's Investors Service for providing the historical database on ratings. Bushman thanks the Kenan-Flagler Business School for financial support.

Copyright ©, University of Chicago on behalf of the Institute of Professional Accounting, 2008

1. Introduction

Information asymmetries between contracting parties fundamentally impact the design of optimal debt-financing arrangements (e.g., Leland and Pyle [1977], Diamond [1991], Aghion and Bolton [1992], Holmström and Tirole [1997]). For syndicated loan deals, which involve two or more parties lending to a single borrower, information asymmetries can exist between the borrower and lenders as well as among the lenders themselves. The impact of information asymmetries on syndicated loan deals is manifested in formal contractual features such as loan amount, interest rate, loan maturity, covenants, and performance pricing provisions. Beyond such formal contract provisions, the existence of asymmetric information between contracting parties also affects the ownership structure of the loan syndicate itself. In this paper, we empirically investigate how the debtcontracting value of accounting information, a mechanism that decreases information asymmetries, influences both the ownership structure and the explicit contractual structure of syndicated loan deals for publicly traded borrowers.

Focusing first on ownership structure, we investigate how the proportion of a syndicated loan deal retained by the lead arranger of the deal is affected by the informativeness of the borrower's accounting information for inferring credit quality, after controlling for the *direct use* of accounting information in the formal contract. We hypothesize that, as the ability of accounting numbers to capture deterioration in credit quality on a timely basis increases (i.e., debt-contracting value of accounting information increases), lead arrangers hold a smaller proportion of the syndicated loan deal.¹ When the debt-contracting value of accounting information increases, information asymmetries between the lead arranger and other syndicate participants decrease. Thus, potential adverse selection and moral hazard problems within the syndicate are mitigated and participants reduce demands for the lead arranger to hold a larger stake of the loan for incentive purposes.

Further, we explore the extent to which the relation between the debtcontracting value of accounting information and the proportion of the loan retained by the lead arranger is conditioned by the presence of alternative disciplining mechanisms. We consider three characteristics that could potentially change the relative influence of accounting information on the proportion retained by the lead arranger: (1) the existence of a credit rating by an independent rating agency, (2) the reputation of the lead arranger, and (3) whether or not the lead arranger has served as a lead arranger for the same borrower on a previous syndicated loan deal. We hypothesize that accounting information with high debt-contracting value is relatively more

¹ As discussed in more detail below, our paper extends existing research that examines the determinants of loan syndicate structure. Important contributions include Simons [1993], Dennis and Mullineaux [2000], Lee and Mullineaux [2004], Jones, Lang, and Nigro [2005], François and Missonier-Piera [2007], and Sufi [2007].

important in reducing the proportion of the loan retained by the lead when the borrower is not rated, when the lead arranger does not have an established reputation, and when the lead arranger has not previously served as a lead arranger for the same borrower. We provide empirical evidence consistent with these predictions.

Existing theories explore the role of information asymmetry in explaining key differences between relationship oriented, single-lender bank loans where the lender holds the whole loan and arm's length loans (e.g., public bonds) where an underwriter retains none of the loan (e.g., Diamond [1991], Boot and Thakor [2000]). Syndicated loans lie on a continuum between these two extremes. Syndicated loan deals are characterized by the existence of a lead arranger who establishes a relationship with the borrowing firm, negotiates terms of the contract, and organizes a syndicate of participant lenders who each fund part of the loan. The exclusive relationship between lead arrangers and borrowers, and the unobservability of the lead's due diligence and monitoring efforts, create potentially severe adverse selection and moral hazard problems that must be considered in optimal syndicate design.²

Lead arrangers, by virtue of their exclusive relationship with the borrower, may ex ante possess private information about the borrower not known to other syndicate members, creating a demand for lead arrangers to hold a proportion of the loan that is increasing in the severity of the adverse selection problem (e.g., Leland and Pyle [1977]). Also, loan participants rely on lead arrangers to perform due diligence on the borrower before the loan is made. Such due diligence efforts are largely unobservable to syndicate participants, creating potential for shirking by the lead arranger. Finally, there is a need to monitor borrowers on an ongoing basis after a loan deal has closed. The multiparty nature of loan syndicates creates a demand for delegation of monitoring activities to mitigate costly duplication of monitoring efforts and potential free-riding problems (e.g., Holmström [1982], Diamond [1984]). While delegating direct monitoring to lead arrangers seems natural, the unobservability of monitoring effort creates potential for shirking, resulting in a demand for lead arrangers to retain a higher percentage of the loan.³

² The relative importance of moral hazard versus adverse selection problems in shaping syndicated loans is an open question. For example, Sufi [2007] provides evidence consistent with moral hazard as the dominant issue, while Wittenberg-Moerman [2006a] supports an adverse selection story. As discussed later, we follow Sufi's [2007] approach to parsing out the two stories.

³ It seems quite plausible that accounting reports can reduce ex ante adverse selection by enabling syndicate participants to assess whether the lead bank is misleading them about the true credit quality of the firm, and to reduce concerns with the lead shirking on ex ante due diligence activities. The key is that, before the contract is signed, the participants have the power to directly act on accounting information by demanding changes in terms or backing out of the deal. However, while accounting information plays an important ex post role through its explicit inclusion in the contract (i.e., covenants, performance pricing), it is less obvious how such

While retaining a larger portion of a loan can serve to mitigate such incentive issues (e.g., Leland and Pyle [1977], Holmström and Tirole [1997]), it is potentially costly as increased exposure to a single borrower restricts diversification of the lead arranger's loan portfolio. Thus, in equilibrium, the syndicate ownership structure reflects an optimal level of loan retention by the lead arranger that trades off costs and benefits given the information environment and the set of alternative mechanisms available to deal with information asymmetries.

A key determinant of the information environment is the informativeness of publicly reported accounting data of the borrower.⁴ Financial accounting systems provide a credible, low cost information set that forms the foundation of the firm-specific information set available for addressing agency problems. Publicly available accounting information contained in general purpose financial statements can mitigate adverse selection and moral hazard through a general transparency channel that is distinct from the formal contracting channel. More public transparency relative to a borrower's credit quality can allow syndicate participants without a privileged relationship to more effectively assess a borrower's credit quality via arm's length monitoring activities, mitigating both adverse selection and moral hazard problems.⁵ In addition, contracting theory supports a crucial role for informative, verifiable performance measures in formal contracting arrangements (e.g., Jensen and Meckling [1976], Holmström [1979], Watts and Zimmerman [1986], Leftwich [1983]).

We conceptualize the debt-contracting value of accounting information as the ability of publicly reported accounting data to predict deteriorations in the credit quality of a borrower on a timely basis. Inherent limitations

information can overcome ex post moral hazard problems deriving from the unobservability of monitoring efforts. In particular, even if the participants are able to observe the borrower taking a value-reducing decision not caught by the lead, if no explicit contractual term is violated, it is not clear what actions can be taken. We discuss this further in section 3 of the paper.

⁴ In this paper we focus on publicly traded firms whose accounting statements are publicly reported. It is of course plausible that, for private firms, the private dissemination of accounting reports to syndicate members would serve a similar role in reducing information asymmetry among loan participants, and thus impact the proportion of loans retained by lead arrangers. However, we do not have access to the financial information of private borrowers in Dealscan, and even if we did, we would not know the dissemination patterns of such private accounting reports to potential syndicate participants. It is also the case that little is known about the quality of publicly reported accounting information versus that of private company accounting statements (see, e.g., Ball and Shivakumar [2005]). We note that Sufi [2007] documents that the proportion of loans retained by lead arrangers is generally higher for private companies relative to public companies, suggesting significant differences in transparency. We leave the question of the role of accounting information for private borrowers to future research.

⁵ Arm's length monitoring via general purpose financial statements is recognized as a key element in supporting the existence of liquid, public capital markets (e.g., Ball [2001], Black [2001], Bushman and Smith [2001], Watts and Zimmerman [1986]). For example, market monitoring is posited as a key mechanism for the prudential regulation of banks as evidenced by Pillar 3 (Market Discipline) of the Basle II Accord (BIS [2003]).

in the ability of a borrower's accounting numbers to reflect reductions in credit quality in a timely fashion limit its role in mitigating adverse selection and moral hazard problems within the syndicate.⁶ We estimate a direct proxy for the debt-contracting value of accounting by exploiting observable changes in credit quality. Our primary measure is a goodness-of-fit statistic from a probit model where credit ratings downgrades are modeled as a function of lagged, seasonally adjusted changes in accounting earnings. We abstract away from issues of earnings management by estimating the debtcontracting value of accounting at the industry level. That is, we attempt to measure a property of a firm's accounting information that is beyond the choice of managers and derives from differences in the inherent ability of the economy-wide accounting regime to capture changes in economic fundamentals on a timely basis across industries. This simple and intuitive variable measures the extent to which reported earnings as a stand-alone measure capture timely information concerning future deteriorations in credit quality.7

While it is beyond the scope of this paper to comprehensively examine the whole range of accounting properties that may support debt-contracting value, we do consider two other prominent properties of accounting information. Specifically, we compute timely loss recognition (Basu [1997]) and estimate the extent to which current earnings capture the information set underlying contemporaneous changes in stock price using the R^2 from a regression of stock returns on current earnings.^{8,9}

We document that, as hypothesized, the proportion of the loan retained by the lead arranger is a decreasing function of the debt-contracting value of accounting information. In the empirical specification, we control for the existence of debt covenants and performance pricing provisions based on accounting variables (i.e., the formal contracting role of accounting); a wide range of firm-specific, loan-specific, and industry-specific characteristics;

⁶ In a different vein, Bharath, Sunder, and Sunder [2008] examine how the magnitude of operating accruals impacts interest rate spreads, loan maturity, and the use of collateral in the private and public debt markets.

⁷ As discussed below, to alleviate concerns about our use of an industry-level debt-contracting value measure, we perform a battery of robustness tests. We also estimate an extended debt-contracting value model that incorporates a wider range of accounting measures, and where possible we examine firm-specific measures of quality.

⁸ Timely accounting recognition of economic losses is a commonly used proxy for accounting conservatism. The potentially important role played by conservative accounting in debt contracting is well known in the literature (e.g., Watts and Zimmerman [1986], Ball [2001], Watts [2003a, b]). Several papers examine efficiency gains from accounting conservatism and/or timely loss recognition in debt contracts (e.g., Ahmed et al. [2002], Zhang [2004], Ball, Robin, and Sadka [2008], Beatty, Weber, and Yu [2006], Wittenberg-Moerman [2006b], Vasvari [2006]).

⁹ We examine earnings timeliness in a debt contracting setting. In the context of shareholder governance, Bushman et al. [2004] investigate how board structure, equity incentives of directors, ownership concentration, and executive compensation vary with earnings timeliness measured using a variant of this R^2 measure.

and key characteristics of the lead arranger. We also document that the negative relation between the proportion retained by the lead arranger and the debt-contracting value of accounting information is larger when the borrower is not rated, when the lead arranger does not have an established reputation, and when the lead arranger has not previously served as a lead arranger for the same borrower.

Our analysis of syndicate ownership structure extends the literature in several important ways. First, we connect the existence of unresolved information asymmetries with direct, intuitive measures of the debt-contracting value of accounting information. This allows us to provide more textured evidence on the central role of accounting information in the design of loan syndicates. Second, we distinguish an important transparency channel through which the inherent properties of general purpose financial statements can contribute to the efficiency of debt contracting, as distinct from a formal contracting channel, such as financial covenants, where lenders can choose their own accounting methods by modifying existing generally accepted accounting principles (GAAP) rules. Third, we document that the relation between the debt-contracting value of accounting information and ownership structure of the syndicate is conditioned by the presence of alternative disciplining mechanisms.

In our final analysis, we extend the literature on the direct contracting role of accounting information by examining how the debt-contracting value of accounting data influences the choice of the performance measure used in performance pricing provisions. Performance pricing provisions index the interest rate charged on a syndicated loan to changes in a contractually chosen measure of borrower performance. These provisions are generally indexed by a single performance measure and are distinguished by whether the measure selected is the borrower's current credit rating or an accounting-based financial ratio. The constraint to one measure implicit in performance pricing creates a tension between the timeliness and informativeness of the performance measure with respect to credit quality, necessitating a trade-off. We hypothesize and document that, conditional on choosing to include a performance pricing provision, syndicates are more likely to choose the timeliness inherent in an accounting ratio over the superior informativeness of credit ratings as the debt-contracting value of a borrower's accounting information increases. We thus establish a direct connection between the choice of performance measure included in the explicit debt contract and the debt-contracting value of accounting.

The paper is organized as follows. Section 2 presents a short institutional background on the syndicated loan market. Section 3 elaborates on the conceptual framework underlying our hypotheses and the relation of our analysis to the existing literature. Section 4 describes our estimation of the debt-contracting value of accounting information, while section 5 presents the empirical analysis of the syndicate ownership structure. We present the analysis on the choice of performance measures in performance pricing provisions and the related results in section 6. Section 7 concludes the paper.

2. Syndicated Loan Market

Syndicated lending is a significant source of corporate financing and has recently generated more underwriting revenue than either the equity or the bond market (Altunbas, Gadanecz, and Kara [2006]). Syndicated loans are loans provided to a borrower by two or more banks. Each member of the lending syndicate has a separate claim on the borrower (not necessarily in equal amounts), but is governed by the same loan agreement. Syndicated loans are usually structured in packages (or deals) of multiple facilities (or loans) with different maturities and repayment schedules.

Syndication helps lenders to avoid capital requirement constraints imposed by regulators (Simons [1993]) as well as limit excessive exposure to individual borrowers. Members of the syndicate can be either senior syndicate bank members (such as lead arrangers, lead managers, and agents) or junior bank participants.¹⁰ The senior banks (hereafter lead arrangers) gather information about the borrower, search for junior bank participants, and coordinate all negotiations. Once a syndicated loan deal is executed, lead arrangers are responsible for monitoring the compliance of the borrower with the contractual terms and the quality of the collateral if the syndicated loan deal is secured, and typically act as administrative agents on behalf of the junior syndicate participants (i.e., collect payments, renegotiate the contract, etc.). Senior banks usually have strong lending relations with the borrowers and receive significant upfront fees in exchange for arranging the syndication deal and taking the underwriting risk. Junior banks typically earn only the interest rate margin, do not have origination capability, and are interested in generating future business from the borrower such as treasury management or advisory work (Altunbas, Gadanecz, and Kara [2006]).

In this paper, we investigate syndicated loan agreements at the origination date (i.e., the primary market) as provided by the Loan Pricing Corporation through the Dealscan database. This database captures a significant proportion of syndicated arrangements, both in the United States and abroad, and provides detailed information on the terms of each loan contract. The terms usually include a set of nonprice terms such as loan maturity, loan size, collateral requirements and covenant restrictions, as well as a set of price terms such as interest rates and fees. Interest rates charged on syndicated loans are expressed as a spread quoted in basis points over a floating benchmark which can be the London Interbank Offered Rate (LIBOR) or another risk-free rate equivalent (e.g., prime rates or T-bills). These variable interest rates are reset every one, two, three, or six months to reflect changes in the benchmark rate.

¹⁰ We classify senior syndicate members as the banks that receive the following descriptions in the Dealscan database: lead bank, lead manager, lead agent, lead arranger, agent, arranger, and book runner.

More recently, the pricing of syndicated loans has become more flexible by adding performance pricing features, which represent a significant shift from the more established use of financial covenants. In contracts that include only financial covenants, the lenders can increase the interest rates only when financial performance deteriorates such that a covenant violation occurs (Smith and Warner [1979], Dichev and Skinner [2002]). If, however, the financial performance improves over the life of the loan, financial covenants do not trigger lower interest rates. This asymmetric response of interest rates to borrower performance is mitigated by performance pricing provisions in the contract. Asquith, Beatty, and Weber [2005] document the role of an important feature of these provisions, the choice of interestincreasing or interest-decreasing pricing grids in resolving adverse selection and moral hazard problems.¹¹

Performance pricing features tie loan interest rates to a borrower-specific performance measure that is typically based on either a financial ratio or a credit rating. Changes in financial ratios or credit ratings automatically adjust the interest rates according to a specific pricing grid agreed upon at the time when the syndicated loans are negotiated. For this reason, performance pricing features can decrease postcontracting costs. On the one hand, lenders benefit from timely protection against sudden drops in the credit quality of the borrower and avoid costly and lengthy debt contract renegotiations. On the other hand, borrowers are rewarded in a timely manner for improvements in their credit quality without the need to incur additional refinancing costs.

3. Conceptual Framework and Related Literature

The framework underlying our empirical tests is rooted in the agency problems of adverse selection and moral hazard created by the existence of information asymmetries among contracting parties.

Consistent with adverse selection (Leland and Pyle [1977]), possession of private, precontracting information can result in the lead arranger holding a substantial stake in the loan. In addition, a moral hazard problem (Diamond [1984], Gorton and Pennacchi [1995], Holmström and Tirole [1997]) may exist between a lender charged with monitoring responsibilities (e.g., lead arranger) and uninformed lenders (e.g., syndicate participants). To create incentives for monitoring, a lead arranger charged with monitoring responsibilities must retain a financial stake in the loan due to the inability of syndicate participants to directly observe the level of monitoring activities.

¹¹ Interest-increasing pricing allows the lender to automatically increase the interest rate over the life of the loan if the creditworthiness of the borrower declines (initial interest rates are low). Interest-decreasing pricing allows the borrower to pay lower interest rates when its credit quality improves (initial interest rates are high). For more details see Asquith, Beatty, and Weber [2005].

The share retained is larger as the moral hazard problem increases. However, this is potentially costly as increased exposure to a single borrower restricts diversification of the lead arranger's loan portfolio.¹²

We argue that publicly available accounting information with high debtcontracting value can help mitigate information asymmetries and improve the efficiency of syndicated loan contracts. In particular, we argue that publicly available accounting information contained in general purpose financial statements can mitigate adverse selection and moral hazard through a general transparency channel that is distinct from the formal contracting channel. In a syndicated loan setting, one can identify at least three distinct information asymmetry problems, two that are potentially present at the time the contract is signed (ex ante) and one that operates after the contract signing (ex post). Accounting information that operates through this general transparency channel is potentially important in alleviating both the ex ante and ex post information problems.

First, consider ex ante contracting issues. Prior to contract signing, lead arrangers, by virtue of their exclusive relationship with the borrower, possess private information about the borrower not known to other syndicate members. In this case, syndicate participants without privileged access to the borrower's inside information will be concerned that a privately informed lead arranger may attempt to sell them larger proportions of low quality loans while keeping higher proportions of high quality loans for themselves. Also, participants rely on lead arrangers to perform due diligence on the borrower prior to loan initiation.¹³ Such due diligence efforts are costly and typically unobservable to syndicate participants, creating potential for shirking by the lead arranger. It seems quite plausible that accounting reports can reduce ex ante adverse selection by enabling participants to assess whether the bank is misleading them about the true credit quality of the firm and can reduce concerns about the lead arranger shirking on ex ante due diligence activities by enabling participants to assess credit quality for themselves. The key is that before the contract is signed, the participants have the power to directly act on accounting information by demanding changes in terms or backing out of the deal.

Second, while accounting information plays an important ex post role through its explicit inclusion in the contract (i.e., covenants, performance pricing), it is less obvious how such information can overcome potential shirking by the lead arranger ex post. Even if the syndicate participants can observe the borrower engaging in a value-reducing activity ignored by

¹² Ivashina [2007] empirically analyzes the trade-off between incentive problems and diversification in determining the optimal proportion of the loan retained.

¹³ Lead arrangers might negotiate higher upfront fees, which can reduce their incentives to perform due diligence ex ante and monitor the borrower ex post. Unfortunately, we cannot investigate this hypothesis because the database does not provide proper coverage of upfront fees received by lead banks. In our sample, less than 1% of the deals have information on upfront arrangement fees.

the lead arranger, it is not clear what actions they can take if no explicit contractual term is violated.

For example, suppose a bank can exert effort to prevent a firm from investing in a bad project, but the lead bank shirks and does not intervene. The question is, how does knowing the borrower made a value decreasing investment help participant lenders from losing additional money? Public accounting information may provide a timely signal that the borrower has embarked on the first stage of a multistage investment project that is bad for the lenders, but that does not involve a covenant violation. While the participant lenders have an opportunity to confront the borrower in an effort to stop further investment, it is not clear that they have the power to intervene.¹⁴ Thus, while it appears that the case is stronger for an important role for accounting in the ex ante adverse selection and moral hazard cases than for the ex post shirking case, we have no real way of distinguishing the ex ante cases from the ex post cases in the data.

However, we do attempt to distinguish between adverse selection and moral hazard. Sufi [2007] argues that, under a moral hazard interpretation, a lead arranger on a current syndicated loan who has previously served as a lead arranger for the same borrower on a previous loan has already expended significant effort to learn about the borrower, and thus will require fewer incentives to exert monitoring effort (i.e., the percentage retained in the loan is *lower*). For example, a first-time lead arranger for a particular borrower has to exert relatively more effort to monitor that borrower than a repeat lead arranger. However, if the existence of a previous lending relationship instead captures the private information advantage of the lead arranger, we would expect to see the percentage retained by the lead to be higher when a previous lending arrangement exists. Sufi [2007] documents that lead arrangers without previous lending relationships hold a larger percentage of the loan, consistent with moral hazard. We extend Sufi [2007] by investigating whether the moral hazard problem captured by a lead arranger with no prior relationship with the borrower is mitigated when the debt-contracting value of accounting increases.

We further hypothesize that the relation between the debt-contracting value of accounting and the proportion of the loan retained by the lead arranger is conditioned by other important aspects of the economic environment. We conjecture that the debt-contracting value of accounting is relatively more important in reducing the fraction of the loan retained by the lead arranger when the borrower is not rated. This is consistent with credit rating agencies providing independent information that reduces information asymmetries between lead banks and syndicate participants and

¹⁴ Many syndicated loan contracts include in the definition of default the occurrence of a loosely defined material adverse change in the financial condition of the borrower. While an event of default allows the lender participants to intervene, it is not clear how effective material adverse change clauses are in allowing participant intervention in the absence of a covenant violation or other explicit contractual violation.

can substitute for the debt-contracting value of accounting information in the design of syndicate structure. We also conjecture that there is a similar substitute effect relative to lead arranger reputation. Pichler and Wilhelm [2001] and Sufi [2007], among others, argue that lead arranger reputation can serve as an effective mechanism in reducing ex ante and ex post moral hazard. Thus, debt-contracting value can reduce the loan fraction retained by lead arrangers relatively more at the margin for lead arrangers with less established reputations in the syndicated loan marketplace.

Turning to the role of accounting information after the loan is made, we focus on a subset of publicly traded firms that include a performance pricing provision in the loan contract.^{15, 16} We argue that the optimal performance measure choice in performance pricing involves a trade-off between the informativeness of a performance measure with respect to the credit quality of the borrower and its *timeliness* (see also Doyle [2003]).¹⁷ All else equal, a lender prefers to use the performance measure that is most informative about the underlying credit quality of the borrower. However, the timeliness of a performance measure is also clearly important. While it is likely that credit ratings are more informative about credit quality of the borrower than any single accounting ratio, the credit ratings process, by its very nature, must sacrifice a certain degree of timeliness. Before issuing a rating change, analysts from the rating agency meet with the management of the borrower several times while writing comprehensive assessment reports. This due diligence process certainly takes time. We hypothesize that, conditional on choosing to include a performance pricing provision, syndicates are more likely to choose an accounting ratio over credit ratings as the borrower's debt-contracting value of accounting information increases.

4. Debt-Contracting Value of Accounting Information

There is little consensus in the accounting literature on which specific, measurable properties of accounting information enhance its value in debt contracting. There are many such properties that could be considered. For example, some argue that conservative accounting numbers enhance the debt-contracting value of accounting (e.g., Ball [2001], Watts [2003a]),

¹⁵ Asquith, Beatty, and Weber [2005] posit that such performance pricing provisions are used by banks to mitigate both moral hazard and adverse selection between banks and borrowers.

¹⁶ Ivashina [2007] and Panyagometh and Roberts [2002] demonstrate a delegated monitoring role for performance pricing provisions by documenting that the percentage retained by the lead arranger is lower when the loan contains a performance pricing provision.

¹⁷ It is interesting to contrast our study with the compensation literature that studies the choice of performance measures. There, the contract trades off *sensitivity* (how sensitive a measure is to managerial actions) and *precision* (measurement error relative to inferring managerial actions). In compensation contracts, multiple measures can be used, and an important issue is how to weigh the measures in the contract. The constraint to one measure in performance pricing creates a tension between the timeliness of a measure and its informativeness relative to the credit quality of the firm (not relative to the actions of the managers).

while others argue to the contrary (e.g., Schipper [2005], Guay and Verrecchia [2007]). But even accepting the importance of conservatism for debt contracting, many different measures are used in the literature to proxy for the general construct of accounting conservatism (see e.g., Beatty, Weber, and Yu [2006]). Beyond conservatism, the literature considers many other attributes of accounting information that often fall under the rubric of "earnings quality" (e.g., Francis et al. [2004]). In an attempt to parsimoniously deal with the multiplicity of attributes, we exploit observable changes in credit quality to create a direct proxy for the debt-contracting value of accounting information.

Our primary measure of debt-contracting value, *DCV*, is generated by estimating a model of credit ratings downgrades as a function of lagged, seasonally adjusted changes in accounting earnings.¹⁸ This variable measures the ability of innovations in quarterly accounting earnings, as a stand-alone measure, to predict credit quality deteriorations in a timely manner. Specifically, *DCV* is measured as Somers' D, a goodness-of-fit statistic, from the following Probit regression:

$$P(Downgrade_{t,i} = 1)$$

= $f(\alpha_0 + \alpha_1 \Delta E_{t-1,i} + \alpha_2 \Delta E_{t-2,i} + \alpha_3 \Delta E_{t-3,i} + \alpha_4 \Delta E_{t-4,i}),$ (1)

where $Downgrade_{t,i}$ is an indicator variable equal to 1 if firm *i*'s credit rating is downgraded in the current quarter *t* (equal to 0 otherwise), and $\Delta E_{t-k,i}$ is the seasonally adjusted change in quarterly earnings before extraordinary items scaled by total assets in the k^{th} quarter prior to the current quarter, *t*. Somers' D, also known as the accuracy ratio, is a popular statistic that is used to measure the quality of credit-rating systems. Basically, Somers' D measures the extent of concordance between the model-predicted downgrades and the actual downgrades. The higher the Somers' D, the higher the downgrade prediction ability of earnings changes (e.g., Altman and Sabato [2007]).¹⁹

We estimate a separate probit regression for each two-digit industry (our loan sample contains 63 distinct two-digit Standard Industrial Classification (SIC) codes). In sensitivity tests, we also estimate the regressions for threedigit and four-digit SIC industry codes. This approach allows us to abstract away from firm-specific issues of earnings management and increases the power of the estimation given a limited number of rating downgrades. We

 $^{^{18}}$ In sensitivity analyses, we also include upgrades in the estimation of *DCV* and estimate an extended model with additional explanatory variables, and find similar results.

¹⁹ Somers' D is a statistic of association between observed downgrades and model predicted downgrade probabilities and is computed as: $(n_c - n_d)/t$, where *t* is the total number of paired observations with different responses in the sample (i.e., one observation is a downgrade, one is not), n_c is the number of concordant pairs, and n_d is the number of discordant pairs. A pair of observations is said to be concordant (discordant) if the observation with a downgrade event has a larger predicted event probability than the paired observation, which is not a downgrade (for detailed explanations see Somers [1962]).

extract the ratings downgrade data from the Moody's Investors Service historical database over the period 1985–2004. We also investigate whether the log-likelihood test statistic is significant for each of the industry-specific probit models. All models have a statistically significant log-likelihood test, meaning that there is a significantly strong relationship between seasonally adjusted earnings changes and downgrade events.

For completeness, we also consider two other prominent properties of accounting information in our main tests. First, we estimate *TIMELINESS* as the extent to which current earnings capture the information set underlying contemporaneous changes in stock price. *TIMELINESS* is the R^2 from the following regression of stock returns on earnings and change in earnings (a similar design is implemented by Bushman et al. [2004], among others):

$$R_{t,i} = \alpha_0 + \alpha_1 E_{t,i} + \alpha_2 \Delta E_{t,i} + \varepsilon_{t,i}, \qquad (2)$$

where $R_{t,i}$ is the four-month market-adjusted stock return ending one month after the end of fiscal quarter t, $E_{t,i}$ is quarterly earnings before extraordinary items scaled by average total assets, and $\Delta E_{t,i}$ is seasonally adjusted changes in quarterly earnings before extraordinary items scaled by average total assets. As with equation (1), we estimate equation (2) at the two-digit industry level and over the period 1985–2004.

Second, we estimate a measure of timely loss recognition, *TLR*, to capture the asymmetric recognition of losses in earnings. Following Basu [1997], *TLR* is estimated as the coefficient on the interaction variable, α_3 , computed from the following regression:

$$E_{t,i} = \alpha_0 + \alpha_1 Neg_{t,i} + \alpha_2 R_{t,i} + \alpha_3 (R_{t,i} \cdot Neg_{t,i}) + \varepsilon_{t,i}, \qquad (3)$$

where $E_{t,i}$ is quarterly earnings before extraordinary items scaled by average total assets, $R_{t,i}$ is market-adjusted, quarterly returns, and $Neg_{t,i}$ is an indicator variable equal to one if the market-adjusted return is negative and zero otherwise. For consistency, equation (3) is also estimated at the two-digit industry level and over the period 1985–2004. *TLR* is expected to be positive and increasing as firms in the industry implement more conservative accounting choices.

In unreported analyses, we also use an asymmetric timeliness measure based on a model implemented by Ball and Shivakumar [2005] and obtain qualitatively similar results.²⁰ In addition, we use industry-specific averages of standardized non-operating accruals (similar to Givoly and Hayn [2000]) and special items as alternative accrual-based conservatism measures.²¹ Our results are robust to both of these measures.

 $^{^{20}}$ We run regressions of accrual levels on cash flows from operations and an indicator variable that takes the value one if the cash flows are negative. Similarly, the asymmetric timeliness measure is the coefficient of the interaction variable.

²¹ Nonoperating accruals are computed as the difference between total accruals and operating accruals (see Givoly and Hayn [2000] for more details).

5. Empirical Analysis of Syndicate Ownership Structure

In this section, we present our empirical analysis of how the debtcontracting value of accounting information influences the fraction of a syndicated loan deal retained by the lead arranger. Section 5.1 outlines the empirical design, section 5.2 discusses sample selection and descriptive statistics, section 5.3 presents the main results, and section 5.4 addresses robustness issues.

5.1 EMPIRICAL DESIGN

Our primary dependent variable is the fraction of the entire deal retained by the lead arranger in the syndicate, *LEAD_OWN*. If there is more than one lead arranger, we follow Sufi [2007] and compute the average share retained. We estimate ordinary least squares (OLS) regression models at the deal (or package of loans) level that relate the lead arranger ownership to our debtcontracting value of accounting information proxies as well as an extensive set of control variables:

$$LEAD_OWN = \alpha_0 + \alpha_1 DCV + \alpha_2 TIMELINESS + \alpha_3 TLR + \sum_{k=1}^{K} \beta_k Controls + \varepsilon,$$
(4)

where *DCV*, *TIMELINESS*, and *TLR* are the debt-contracting value of accounting information measures described in the prior section.

Beyond the debt-contracting value of accounting information, three attributes play a central role in our empirical design: (1) whether or not a borrower or the loan is rated by a credit rating agency, (2) whether or not the lead arranger has served as a lead arranger on a previous loan for the borrower, and (3) the reputation of the lead arranger in the syndicated loan market. First, we define UNRATED as an indicator variable equal to one if the firm or the loan is not rated and equal to zero otherwise.²² This is a key variable since we view credit rating agencies as an alternative source of information that decreases information asymmetries between syndicate members. Sufi [2007] argues that unrated firms are opaque and finds that lead arrangers are required to hold more of the syndicated loan deal. Second, we define NO_LEAD_PRIOR as an indicator variable equal to one if the current lead arranger was not a lead arranger for the same borrower in a previous deal and equal to zero otherwise. Lead banks with no previous loan deals with the current borrower must invest significantly more effort to learn about and monitor the borrower. Thus, lead arrangers must hold a larger share in the deal, which creates an incentive for them to expend the additional effort and resources in monitoring a relatively unfamiliar borrower (Sufi [2007]). Finally, we define LEAD_REPUTATION as an indicator

²² We search both Dealscan and the Moody's Investor Service database for ratings availability.

variable equal to one if the lead arranger is classified as a top 25 lead arranger in the U.S. syndicated loan market during the year the deal is signed and equal to zero otherwise.²³

We also consider a number of other deal-specific control variables in multivariate tests (a description of all variables is included in the appendix). If a deal has multiple loans, we select characteristics of the largest loan facility in the deal. DEAL_SIZE, defined as the logarithm of the total dollar value of the deal, is a proxy for the overall syndicate risk exposure. Since larger deals are expected to be financed by a larger number of syndicate members (due to capital requirement constraints or to limit risk exposure to one borrower), we expect lead arrangers to retain a smaller ownership percentage. LOAN_SPREAD is the all-in-spread drawn (in basis points over LIBOR) from Dealscan. Dealscan defines the all-in-spread drawn as the total annual spread paid for each dollar drawn down under the loan commitment (including fees). We include the spread of the largest loan in the deal to control for borrower-specific risk characteristics that are not captured by other variables. LOAN_MATURITY is the number of years to loan maturity. SECURED is an indicator variable equal to one if the loan is secured with collateral and equal to zero otherwise. REVOLVER is an indicator variable equal to one if the loan is revolving and equal to zero otherwise.²⁴

Three variables are included to control for the direct contracting role of accounting information. *GEN_COVENANTS* and *FIN_COVENANTS* are computed as the number of general and financial covenants, respectively, and *PP_INDICATOR* is an indicator variable equal to one if the loan contains a performance pricing provision and equal to zero otherwise.²⁵

We further control for several key borrower-specific variables. LOG_PREVIOUS, computed as the logarithm of one plus the number of previous syndicated loans taken by the borrower, is a proxy for the reputation of the borrower in accessing the syndicated loan market. The remaining variables are defined as follows: PROFITABILITY is operating income scaled by average total assets at the time of the deal, INTEREST_COVERAGE is the sum of earnings before extraordinary items and interest expense scaled by interest expense, FIRM_SIZE is the logarithm of the book value of total assets

²³ We retrieve historical league tables from SDC Platinum (Thompson Financial) with the top 25 managing underwriters by the size of deals closed in the U.S. syndicated loan market.

²⁴ The *REVOLVER* variable might not fully capture differences between revolving and term loans in the multivariate regressions. Therefore, in an unreported sensitivity test, we re-estimate the regressions using only revolvers (approximately 84% of our sample). The results are not affected.

²⁵ Financial and general covenants available in Dealscan are presented in Vasvari [2006]. Financial covenants are promises not to allow certain balance sheet or income statement items or ratios to fall below (or above) an agreed upon level (e.g., net worth, current ratio, interest coverage, debt to equity). General covenants are standard assurances and undertakings that the syndicate obtains from the borrower (e.g., loan must be repaid out of excess cash flows, debt, asset sales, equity issues or insurance proceeds (sweeps), restrictions on dividend payments, etc.).

(we expect lead arrangers to hold a relatively smaller share of the deal when the borrowing firm is larger), and *LEVERAGE* is the sum of the borrower's debt in current liabilities and total long-term debt scaled by total assets. Finally, we control for year fixed effects to capture structural changes in the syndicated loan market liquidity over time.

5.2 SAMPLE SELECTION AND DESCRIPTIVE STATISTICS

We collect our sample from Dealscan (provided by the Loan Pricing Corporation), which contains data on syndicated loan agreements at the time of their origination. The syndicated loan data are gathered from the Securities and Exchange Commission (SEC) (13Ds, 14Ds, 13Es, 10-Ks, and 8-Ks) or through Loan Pricing Corporation's relationships with major banks active in the syndicated loan market. The loans (or facilities) are grouped in deals (or packages of loans) when borrowing firms enter into multiple agreements at the same time. We perform our analysis at the deal level, as opposed to the loan level, because syndicated loan contracts are drafted at the deal level and all lenders participate in deal tranches collectively, not independently (e.g., Sufi [2007]).

We apply several filters to the data to obtain the final loan sample for the syndicate ownership analysis (see table 1, panel A). First, we select all loans for which the U.S. borrower could be manually matched to Compustat based on company name, industry membership, and geographical location. Second, we exclude financial companies and require data availability on percentage holdings of each syndicate member as well as loan-specific and borrower-specific control variables. Finally, we eliminate all sole lender deals to identify a clean sample of syndicated loans. Our final sample contains 4,140 deals made to 1,915 borrowing firms in 1992–2004.

Table 1, panel B presents descriptive statistics. The mean *LEAD_OWN* is 0.25, which is comparable with the value reported in Sufi [2007]. The mean (median) *DCV* is 0.356 (0.339) and exhibits considerable variation across industries.²⁶ *TLR* is positive across all quartiles, suggesting that the accounting choices implemented by borrowers are, on average, conservative at the industry level. The average syndicated deal size is approximately \$450 million. The largest loan in a deal has an average maturity of approximately three and one-half years and a spread of 141 basis points above a benchmark risk-free rate. In our sample, 40% of the loans are secured and 84% are revolving. On average, the deals have about one financial and two general covenants attached. Performance pricing provisions are included in 61% of the deals. The median borrower has four syndicated loans prior to the

 $^{^{26}}$ To put this in perspective, a Somers' D of 0.356 means that the model is 35.60% more accurate in predicting rating downgrades than a random guess. That is, using the predicted values from the probit model, one can correctly distinguish between a downgrade observation (higher predicted value) and a nondowngrade observation (lower predicted value) 67.8% of the time, or 50% * 1.356. Somers' D ranges from -1 to +1, so a value of zero translates into a 50% chance of correctly classifying the observations, or essentially a coin toss.

current deal and is, on average, relatively large with \$4.2 billion in total assets, which is consistent with other studies that use Dealscan data. The average borrower has a 3.0% return on assets and a leverage ratio of 0.34.

Table 1, panel C presents Pearson correlation statistics among the dependent variable and selected independent variables used in the multivariate tests. Of particular interest is the correlation of 0.408 between *DCV* and *TIMELINESS*, which is consistent with a high degree of debt and equity markets integration. However, in the next section, multivariate tests show that *DCV* captures a more significant portion of the variation in our dependent variables than *TIMELINESS*. Finally, we note that the correlation between *TLR* and *DCV* is positive and statistically significant.

TAB	LE 1
-----	------

Syndicate Ownershi	ip Sample Selection	, Descriptive Stat	tistics, and C	Correlatio	m Matrix	
Panel A: Description of syndi	icate ownership s	ample selection	n procedur	e		
Syndicate Ownership Sample	Selection (Deal	Level)	•	Loans	Deals	Firms
-Sample of syndicated loan of	contracts matche	d to Compusta	t :	33,375	24,441	6,243
-Sample after excluding fina	incial firms			29,282	21,061	5,468
 Sample after requiring avail variables 	lability of loan ar	nd borrower-spo	ecific	17,819	12,483	4,002
- Sample after requiring data arrangers and participants	on percentage o	ownership of le	ad	-	5,907	2,788
- Sample after removing sole	lender deals				4,140	1,915
Panel B: Sample distribution	of model variab	les for syndicat	e ownershi	p sampl	e	
	N	Mean	25%		Median	75%
Dependent variables						
LEAD_OWN	4,140	0.25	0.10		0.19	0.36
LENDER_HERF	4,140	0.21	0.08		0.15	0.31
Information variables						
DCV	4,140	0.356	0.271		0.339	0.409
TIMELINESS (%)	4,140	1.287	0.533		0.992	1.747
TLR	4,140	0.382	0.266		0.332	0.476
Loan-specific variables						
UNRATED	4,140	0.36	_		_	_
NO_LEAD_PRIOR	4,140	0.38	_		_	_
LEAD_REPUTATION	4,140	0.63	-		_	_
DEAL_SIZE (\$ million)	4,140	458	75		185	425
LOAN_SPREAD (bps)	4,140	141	55		113	200
LOAN_MATURITY	4,140	3.55	2.00		3.01	5.00
SECURED	4.140	0.40	_		_	_
REVOLVER	4.140	0.84	_		_	_
GEN_COVENANTS	4.140	2.59	0		2	4
FIN_COVENANTS	4,140	1.65	0		2	3
PP_INDICATOR	4,140	0.61	_		_	_
Borrower-specific variables						

5.65

0.03

6.16

4,212

0.34

4.140

4,140

4,140

4,140

4,140

NUM_PREVIOUS_LOANS

INTEREST_COVERAGE

FIRM_SIZE (\$ million)

PROFITABILITY

LEVERAGE

(Continued)

8

0.07

5.59

2.782

0.43

4

0.04

2.76

756

0.32

2

0.01

1.29

270

0.21

Panel C: Pearson con	rrelatio	n matrix e	of selecte	ed model	variables	6				
	LENDER HERF	DCV	TIMELINESS	TLR	UNRATED	NO_LEAD_PRIOR	LEAD_REPUTATION	PROFITABILITY	LEVERAGE	INTEREST_COVERAGE
LEAD_OWN	0.789^{*}	-0.187^{*}	-0.056^{*}	-0.209^{*}	0.448^{*}	0.283*	-0.445^{*}	-0.112^{*}	-0.065^{*}	0.011
LENDER_HERF		-0.279^{*}	-0.105^{*}	-0.278^*	0.430^{*}	0.288^{*}	-0.339^{*}	-0.145^{*}	-0.046^*	0.004
DCV			0.408^{*}	0.157^{*}	-0.056^{*}	-0.057^{*}	0.110^{*}	0.057^{*}	-0.015	0.014
TIMELINESS				-0.090	0.009^{*}	-0.010	0.024	0.046^{*}	0.001	0.017
TLR					-0.100^{*}	-0.066^{*}	0.179^{*}	0.074^{*}	-0.054^{*}	0.065^{*}
UNRATED						0.214^{*}	-0.384^{*}	0.040^{*}	-0.215^{*}	0.116^{*}
NO_LEAD_PRIOR							-0.237^{*}	-0.041^{*}	-0.074^{*}	0.046^{*}
LEAD_REPUTATION								-0.066^{*}	0.019	0.009
PROFITABILITY									-0.353^{*}	0.345^{*}
LEVERAGE										-0.360^{*}

TABLE	1 - Continued

LEAD_OWN is the fraction of the deal owned by the lead arranger. LENDER_HERF is the sum of the squared percentage ownership of each lender in the deal syndicate. DCV is a credit market based earnings quality measure computed as the Somers' D association statistic obtained from industry-specific (two-digit SIC codes) Probit regressions that predict credit rating downgrades. The downgrade predictors are the seasonally adjusted quarterly earnings over the prior four quarters. TIMELINESS is an equity market based earnings quality measure computed as the R^2 obtained from industry-specific (two-digit SIC codes) pooled regressions of market-adjusted returns on quarterly earnings levels and seasonally differenced quarterly earnings. TLR is timely-loss recognition measured as the coefficient on negative returns in earnings-returns regressions estimated using quarterly data at the industry level (2 digit SIC codes). UNRATED is an indicator variable equal to 1 if the loan is rated and 0 otherwise. NO_LEAD_PRIOR is an indicator variable equal to 1 if a lead arranger in the current deal was not a lead arranger in a previous deal with the same borrower. LEAD_REPUTATION is an indicator variable equal to 1 if the lead arranger is classified in the top 25 arrangers of syndicated loans in the U.S. in the year when the deal is signed. DEAL_SIZE is the size of the deal (\$ mil.). LOAN_SPREAD is the loan spread (in basis points). LOAN_MATURITY is the number of years to loan maturity. SECURED is an indicator variable equal to 1 if the loan is collateralized and 0 otherwise. REVOLVER is an indicator variable equal to 1 if the loan is a revolving loan and 0 otherwise. GEN_COVENANTS (FIN_COVENANTS) is the number of general (financial) covenants in the contract as reported by Dealscan. PP_INDICATOR is an indicator variable equal to 1 if the loan contract contains a performance pricing provision and 0 otherwise. NUM_PREVIOUS_LOANS is the number of previous syndicated loans taken by the borrower. PROFITABILITY is operating income before depreciation scaled by average total assets. INTEREST_COVERAGE is the borrower's interest coverage ratio defined as the sum of interest expense and income before extraordinary items scaled by interest expense. FIRM_SIZE is the book value of total assets (\$ mil.). LEVERAGE is the book value of debt (sum of debt in current liabilities and total long-term debt) divided by book value of total assets. * indicates significance at the 1% level based on a two-tailed test.

5.3 RESULTS

Table 2 presents the results of our analysis of syndicate ownership structure and the debt-contracting value of accounting information. Following prior literature (e.g., Sufi [2007]), we compute coefficient significance levels using standard errors adjusted for clustering at the borrower level to account for multiple observations for the same firm.²⁷ Consistent with our first hypothesis, we find a negative and significant (at the 1% level) association between *LEAD_OWN* and *DCV* (column 1). This suggests that, as the debt-contracting value of the accounting information improves, information asymmetries are reduced and there is a lower demand for the lead arranger to hold a larger fraction of the deal. The economic magnitude of

²⁷ In a later sensitivity test discussed in Section 5.4 we also cluster the standard errors at the two-digit SIC industry level and at the lead arranger level.

	ð
	- and -
2	it on
ĽΕ	MAG
N B	140
Ē	I_{f}
	atte

			Determin	nants of Debt M	1 onitoring Structure				
	Predicted				Dependent Varia	hble: LEAD_OV	NA		
	Sign		1		2		3		4
DCV	I	-0.083	$(-4.31)^{***}$			-0.077	$(-3.61)^{***}$	-0.086	$(-4.06)^{***}$
TIMELINESS	I			-0.005	$(-1.96)^{**}$	-0.002	(-0.88)	-0.003	(-1.03)
TLR	I							-0.056	$(-3.35)^{***}$
UNRATED		0.027	$(4.23)^{***}$	0.027	$(4.18)^{***}$	0.027	$(4.24)^{***}$	0.027	$(4.25)^{***}$
NO_LEAD_PRIOR		0.025	$(4.78)^{***}$	0.025	$(4.83)^{***}$	0.025	$(4.82)^{***}$	0.025	$(4.80)^{***}$
LEAD_REPUTATION		-0.084	$(-14.01)^{***}$	-0.085	$(-14.08)^{***}$	-0.084	$(-14.03)^{***}$	-0.083	$(-13.82)^{***}$
DEAL_SIZE		-0.056	$(-15.05)^{***}$	-0.056	$(-14.80)^{***}$	-0.056	$(-15.06)^{***}$	-0.056	$(-15.07)^{***}$
LOAN_SPREAD		-0.002	(-0.53)	-0.003	(-0.70)	-0.002	(-0.50)	-0.004	(-0.78)
LOAN_MATURITY		-0.007	$(-4.55)^{***}$	-0.006	$(-4.45)^{***}$	-0.007	$(-4.55)^{***}$	-0.006	$(-4.54)^{***}$
SECURED		0.013	$(2.26)^{**}$	0.014	$(2.40)^{**}$	0.013	$(2.28)^{**}$	0.014	$(2.32)^{**}$
REVOLVER		-0.029	$(-3.63)^{***}$	-0.028	$(-3.60)^{***}$	-0.028	$(-3.60)^{***}$	-0.029	$(-3.67)^{***}$
GEN_COVENANTS		-0.010	$(-1.78)^{*}$	-0.010	$(-1.78)^{*}$	-0.010	$(-1.75)^{*}$	-0.009	$(-1.71)^{*}$
FIN_COVENANTS		-0.012	$(-2.18)^{**}$	-0.011	$(-2.08)^{**}$	-0.012	$(-2.21)^{**}$	-0.011	$(-2.09)^{**}$
PP_INDICATOR		-0.013	$(-1.93)^{*}$	-0.013	$(-2.03)^{**}$	-0.013	$(-1.92)^{*}$	-0.013	$(-1.93)^{*}$
									(Continued)

265

	Predicted			[Dependent Varia	ble: <i>LEAD_O</i> W	N		
	Sign		1		2		3		4
LOG_PREVIOUS		0.005	(1.41)	0.005	(1.47)	0.005	(1.45)	0.005	(1.53)
INTEREST_COVERAGE		-0.144	(0.06)	-0.141 0.002	(-3.74) (0.12)	0.001	(0.08)	-0.149 0.003	(0.27)
FIRM_SIZE		-0.014	$(-4.78)^{***}$	-0.014	$(-4.77)^{***}$	-0.014	$(-4.78)^{***}$	-0.015	$(-4.85)^{***}$
LEVERAGE		-0.028	$(-1.98)^{**}$	-0.026	$(-1.85)^{*}$	-0.028	$(-1.98)^{**}$	-0.031	$(-2.19)^{**}$
Year fixed effects			Yes		Yes		Yes	Υ	es
N Adj. R^2		V. O.	4,140	4 0	l,140 1.563	4 0	,140 .564	4,]	140 567
Dependent variable is <i>LL</i> , association statistic obtained fi quarterly earnings over the pr codes) pooled regressions of coefficient on negative return	<i>ID_OWN</i> , the fitom om industry spinor four quarte market-adjuste s in earnings-re	action of the ecific (2 digit rs. <i>TIMELINE</i> d returns on turns regressi	: deal owned by the SIC codes) Probit re SS is an equity mark quarterly earnings 1 ons estimated using	: lead arranger. sgressions that p ket based earnin levels and seaso quarterly data a	<i>DCV</i> is a credit mirredict credit rating ges quality measure nally differenced of the industry level	arket based earr downgrades. Th computed as th luarterly earning (two-digit SIC c	ings quality measu the downgrade predide $e R^2$ obtained from gs. TLR is timely-lov odes). UNRATED is	re computed as ctors are the sea industry specifi ss recognition n an indicator var	the Somers' D sonally adjusted c (two-digit SIC neasured as the iable equal to 1

TABLE 2-Continued

LEVERACE is the borrower's book value of debt (sum of debt in current liabilities and total long-term debt) divided by book value of total assets. Significance levels of coefficient estimates are based on standard errors adjusted for clustering at the firm level.^{***, ***}, and * indicate significance at the 1%, 5%, and 10% levels, respectively, based on a two-tailed test. borrower. *IEAD-REPUTATION* is an indicator variable equal to 1 if the lead arranger is classified in the top 25 arrangers of syndicated loans in the U.S. in the year when the deal is signed. DEAL_SIZE is the logarithm of the deal size. LOAN SPREAD is the logarithm of loan spread (in basis points). LOAN_MATURITY is the number of years to loan maturity. GEN COVENANTS (FIN COVENANTS) is the number of general (financial) covenants in the loan contract as reported by Dealscan. PP. INDICATOR is an indicator variable equal to 1 if the loan agreement contains a performance pricing provision and 0 otherwise. LOG-PREVIOUS is the log of one plus the number of previous syndicated loans taken by the borrower. PROFITABILITY is the borrower's operating income before depreciation scaled by average total assets. INTEREST COVERAGE is the borrower's interest coverage ratio defined as the sum of interest expense and income before extraordinary items scaled by interest expense. FRM.SIZE is the logarithm of the borrower's book value of total assets. SECURED is an indicator variable equal to 1 if the loan is collateralized and 0 otherwise. REVOLVER is an indicator variable equal to 1 if the loan is a revolving loan and 0 otherwise. if the loan is rated and 0 otherwise. NO.IEAD PRIOR is an indicator variable equal to 1 if a lead arranger in the current deal was not a lead arranger in a previous deal with the same

266

the coefficient is also significant. A one standard deviation change in the DCV (0.12) decreases the lead arranger ownership by approximately 1%, which is a change of about 5.2% of the sample lead ownership median.

Consistent with Sufi [2007], we find that LEAD_OWN is higher when the borrower is unrated (UNRATED = 1) and when the lead arranger did not previously arrange a loan for the borrower (NO_LEAD_PRIOR = 1). Sufi [2007] interprets the positive coefficient on NO_LEAD_PRIOR as consistent with a moral hazard story, where lead arrangers who have not previously arranged loans for a borrower must be given stronger incentives to exert the effort necessary to monitor an unfamiliar borrower.²⁸ We also find that a higher lead arranger reputation (*LEAD_REPUTATION* = 1) decreases LEAD_OWN. Further, DEAL_SIZE, MATURITY, SECURED, and both covenant measures have the expected sign and are significant at conventional levels. Firm-specific controls, such as PROFITABILITY and SIZE, also have the expected sign. In column 2, we find that the coefficient on TIMELINESS is negative and significant at the 5% level. However, in column 3, when both DCV and TIMELINESS are included, TIMELINESS is no longer significant. Finally, in column 4, we introduce all three accounting information measures and find that DCV and *TLR* are negative and significant (at the 1% level), suggesting that each captures different dimensions of the debt-contracting value of accounting

information.

Next, we examine our second hypothesis that the magnitude of the relation between LEAD_OWN and DCV is conditional on whether or not a borrower or the loan is rated by a credit rating agency, whether or not the lead arranger has served as a lead arranger on a previous loan for the borrower, and the reputation of the lead arranger in the syndicated loan market. In table 3, columns 1 to 3, we interact DCV with UNRATED, NO_LEAD_PRIOR, and LEAD_REPUTATION. First, the coefficient on the interaction term DCV* UNRATED is negative and statistically significant, which is consistent with accounting information providing a more important arm's length monitoring role in the absence of delegated monitoring by a credit rating agency. Second, the coefficient on DCV* NO_LEAD_PRIOR is also negative and significant indicating that accounting information is relatively more important in mitigating information asymmetries when a lead arranger does not have a previous lending relationship with a borrower, thus requiring more incentives to exert effort to learn about the borrower. Finally, as expected, we find a positive and significant coefficient on DCV* LEAD_REPUTATION. Lead arrangers with a high reputation at stake require relatively fewer incentives to counteract information asymmetries and so accounting information, as an alternative monitoring mechanism to the lead arranger, becomes relatively less important. In table 3, column 4, we also include the interaction of TLR

²⁸ An alternative explanation is that lead arrangers take more of the deal because they do not have any prior exposure on the borrower.

ດ	C	0	
4	υ	0	

				TABLI	E 3				
			Determine	ants of Debt Mo	onitoring Structure				
	Pred.				Dependent Varia	ble: <i>LEAD_O</i> V	NA		
	Sign		1		2		3		4
DCV	I	-0.079	$(-4.05)^{***}$	-0.042	$(-2.14)^{**}$	-0.062	$(-2.72)^{***}$	-0.071	$(-2.96)^{***}$
DCV* UNRATED	I	-0.016	$(-4.36)^{***}$	-0.052	$(-6.61)^{***}$	-0.049	$(-6.09)^{***}$	-0.052	$(-6.56)^{***}$
DCV*NO_LEAD_PRIOR	Ι			-0.045	$(-5.92)^{***}$	-0.046	$(-5.96)^{***}$	-0.046	$(-5.76)^{***}$
DCV*LEAD_REPUTATION	+					0.045	$(2.73)^{***}$	0.049	$(2.90)^{***}$
TLR								-0.128	$(-3.51)^{***}$
TLR* UNRATED								-0.007	$(-1.61)^{*}$
TLR*NO_LEAD_PRIOR								0.048	$(1.75)^{*}$
TLR*LEAD_REPUTATION								0.097	$(2.61)^{***}$
UNRATED		0.033	$(5.00)^{***}$	0.044	$(6.34)^{***}$	0.043	$(6.11)^{***}$	0.048	$(6.71)^{***}$
NO_LEAD_PRIOR		0.025	$(4.76)^{***}$	0.044	$(7.05)^{***}$	0.044	$(7.05)^{***}$	0.026	$(2.05)^{**}$
LEAD_REPUTATION		-0.085	$(-14.19)^{***}$	-0.085	$(-14.21)^{***}$	-0.101	$(-11.55)^{***}$	-0.136	$(-8.32)^{***}$
DEAL_SIZE		-0.053	$(-13.64)^{***}$	-0.053	$(-13.86)^{***}$	-0.053	$(-13.75)^{***}$	-0.054	$(-13.87)^{***}$
LOAN_SPREAD		-0.001	(-0.31)	-0.001	(-0.31)	-0.001	(-0.28)	-0.003	(-0.57)
LOAN_MATURITY		-0.007	$(-4.63)^{***}$	-0.007	$(-4.72)^{***}$	-0.007	$(-4.74)^{***}$	-0.007	$(-4.71)^{***}$
SECURED		0.015	$(2.46)^{**}$	0.015	$(2.54)^{**}$	0.015	$(2.50)^{**}$	0.014	$(2.47)^{**}$
REVOLVER		-0.030	$(-3.83)^{***}$	-0.030	$(-3.90)^{***}$	-0.030	$(-3.86)^{***}$	-0.030	$(-3.91)^{***}$
									(Continued)

GEN_COVENANTS FIN_COVENANTS PP_INDICATOR	-0.011 -0.012 -0.009	$(-2.01)^{**}$ $(-2.33)^{**}$ (-1.32)	-0.010 -0.013 -0.009	$(-1.90)^{*}$ $(-2.40)^{**}$ (-1.43)	-0.010 -0.013 -0.009	$(-1.91)^{*}$ $(-2.37)^{***}$ (-1.45)	-0.010 -0.012 -0.010	$(-1.89)^{*}$ $(-2.26)^{**}$ (-1.54)
LOG_PREVIOUS PROFITABILITY	0.005 - 0.145	(1.38) $(-3.87)^{***}$	0.003 - 0.143	(0.95) $(-3.89)^{***}$	$0.003 \\ -0.142$	(0.89) $(-3.84)^{***}$	$0.003 \\ -0.145$	(0.93) $(-3.89)^{***}$
INTEREST_COVERAGE FIRM_SIZE	0.005 - 0.013	(0.04) $(-4.44)^{***}$	0.001 - 0.014	(0.10) $(-4.57)^{***}$	0.005 - 0.014	(0.04) $(-4.54)^{***}$	-0.005 -0.014	(0.00) $(-4.58)^{***}$
LEVERAGE	-0.028	$(-1.98)^{**}$	-0.029	$(-2.08)^{**}$	-0.029	$(-2.05)^{**}$	-0.029	$(-2.05)^{**}$
Year fixed effects	~	Yes		Yes 4 140	~	Yes		Yes 140
Adj. R^2	r 0	.1567		0.571	r O	.571	r O	.575
Dependent variable is <i>LEAD_OWN</i> <i>NO.LEAD_PRIOR</i> is an indicator varia as the indicator variable is the downgrade J market downgrades. The downgrade J <i>DCV</i> and <i>UNRATED_DCV</i> * <i>NO.LE4D</i> <i>LEAD_REPUTATION</i> . TLRs it interly-los (wo-digit SIC codes). <i>TLRs.UNRATED_</i> <i>LTRs.LEAD_REPUTATION</i> is an intera spread (in basis points). <i>LOAN_MATTO</i> <i>s</i> an indicator variable equal to 1 if the as reported by Dealscan. <i>PP_IND/CATC</i> of one plus the number of previous sy <i>NTERST_COVERACE</i> is the borrower's boo book value of total assets. Significance 1%, 5%, and 10% levels, respectively, I	i, the fraction of ble equal to 1 it ble equal to 1 it e the lead at 1 it is computed as predictors are the predictors are the <i>PROR</i> is an intu- is recognition m is recognition m is recognition m is in interaction via the nume PRTY is the num $PRTY$ is the nume PRTY is the nume PRTY	of the deal owned b f a lead arranger in it er is a lead arranger in the rer is concas' D associ the seasonally adjust the seasonaly adjust the seasonally adjust the seasonally	y the lead arraph the current deside to 25 arran, the current deside to 25 arran, the constraints of the constraints of the constraints of the constraints of the constraint of the constra	unger. UNRATED is a al was not a lead arra bgens of syndicated lo bbeined from indust lo brained from indust arnings over the pric it NOLEAD_PRIOR. D er erturns in earnings it INOLEAD_PRIOR. D er erturns in earnings it INOLEAD_PRIOR. D er erturns in earnings it is an indicator vari ED is an indicator vari ED is an indicator vari ertest expense and inco ertest expense and inco rest expense and inco rest expense and inco red bort value of debt (o	1 indicator variand indicator varianger in a previound ender the second of the second	the equal to 1 if the use deal with the sam in the year when the heat with the year when the ligit SIC codes) Pro- $DC(s_LURATED is)$ TATTON is an intervaland interval intervaland the low is collateriated and the lowis growision and 0 othare before depreciatedthe before depreciateda provision and 0 othme before depreciateda provision and 0 othme before depreciateda firm level. ***, **,	the loan is rated ne borrower. LE bit regressions to bit regressions to an interaction wein TLR and ween TLR and $SPREAD is thedized and 0 othenancial) coverantrerwise. LOG_{PP}ion scaled by aion scaled by adion $	and 0 otherwise. AD_REPUTATION . DCV is a credit hat predict credit eween DCV and the industry level $VO_{LEAD_{PROR}}$ ogarilatim of loan vise. REVOLVER try in the contract EVIOUS is the log erage total assets. pense. FIRM_SIZE pense. FIRM_SIZE pense. FIRM_SIZE

with UNRATED, NO_LEAD_PRIOR, and LEAD_REPUTATION. As expected, we find that the coefficient on TLR* UNRATED is negative and the coefficient on TLR* LEAD_REPUTATION is positive. However, the coefficient on TLR* NO_LEAD_PRIOR has a sign opposite to what we predict.

5.4 SENSITIVITY ANALYSIS

In table 4, we present a number of sensitivity analyses to ensure the robustness of our results. Panel A presents sensitivity tests using alternative proxies for the debt-contracting value of accounting information and syndicate ownership structure. Panel B provides sensitivity analyses to alleviate concerns associated with our estimation of the debt-contracting value of accounting (the *DCV* variable) at the two-digit industry level, and to deal with potentially correlated omitted variables with respect to the lead arrangers themselves.

5.4.1. Alternative Empirical Proxies. First, we re-estimate the two-digit SIC industry *DCV* model with additional firm-specific accounting variables. We identify variables that are commonly used by rating agencies to establish rating levels (e.g., Moody's Investors Services [2006]) and include their seasonally adjusted changes over the prior year in the following probit regression:²⁹

$$P(Downgrade_{t} = 1)$$

$$= f(\alpha_{1} + \alpha_{2} \Delta E_{t-1} + \alpha_{3} \Delta E_{t-2} + \alpha_{4} \Delta E_{t-3}$$

$$+ \alpha_{5} \Delta E_{t-4} + \alpha_{6} \Delta SALES + \alpha_{7} \Delta SALES_SEG$$

$$+ \alpha_{8} \Delta NOSEG + \alpha_{9} \Delta CASH_DEBT + \alpha_{10} \Delta LEVERAGE).$$
(5)

 $\Delta SALES$ is the change in sales and is a proxy for change in the size of the firm. Large firms have more resources and are more likely to be diversified, which reduces volatility and credit risk. $\Delta SALES_SEG$ is the change in sales of the largest business segment of the company and $\Delta NOSEG$ is the change in the number of business segments (as reported by Compustat Segment Tapes). Both are proxies for diversification of operations, which mitigates the effects of the variation in demand or prices in a given product or market, thus reducing the risk of the borrower. $\Delta CASH_DEBT$ is the change in liquidity strength of the borrowers as a signal of their ability to service the debt. Finally, $\Delta LEVERAGE$ is the change in leverage scaled by the market value of equity one year prior to the rating downgrade. Moody's Investor Services [2006] argue that this measure captures the commitment to manage the balance sheet prudently from the perspective of lenders (rating agencies favor equity financing over debt financing for new projects). Results in table 4,

 $^{^{29}\,\}mathrm{We}$ include changes in each variable because our model predicts rating changes, not levels.

Panel A: Alternative mea	surements of D(JV, TIMELINES	S, and TLR and alter	rnative depende	ent variable.				
				Dependent V	ariable: <i>LEAD_OW</i>	1		Denende	nt Variable:
	Predicted	Extended	d DCV Model	3-Digit S	IC Code DCV	Firm-Level	Timeliness/TLR	TEND	SR_HERF
	Sign		1		2		3		4
DCV	I	-0.063	$(-3.17)^{***}$	-0.042	$(-3.34)^{***}$	-0.071	$(-3.45)^{***}$	-0.071	$(-3.89)^{***}$
TIMELINESS	I	-0.003	(-0.84)	-0.004	(-1.31)	-0.037	(-1.23)	-0.001	(-0.47)
TLR	I	-0.050	$(-3.04)^{***}$	-0.055	$(-3.16)^{***}$	-0.028	$(-2.24)^{**}$	-0.036	$(-2.39)^{**}$
UNRATED		0.031	$(4.82)^{***}$	0.032	$(4.79)^{***}$	0.026	$(3.67)^{***}$	0.021	$(3.62)^{***}$
NO_LEAD_PRIOR		0.022	$(4.24)^{***}$	0.027	$(4.98)^{***}$	0.023	$(4.21)^{***}$	0.022	$(4.57)^{***}$
LEAD_REPUTATION		-0.083	$(-13.89)^{***}$	-0.080	$(-12.63)^{***}$	-0.084	$(-12.56)^{***}$	-0.070	$(-12.60)^{***}$
DEAL_SIZE		-0.057	$(-15.40)^{***}$	-0.058	$(-14.10)^{***}$	-0.055	$(-12.95)^{***}$	-0.052	$(-14.45)^{***}$
LOAN_SPREAD		-0.006	(-1.29)	-0.004	(-0.84)	-0.000	(-0.05)	0.003	(0.74)
LOAN_MATURITY		-0.006	$(-4.10)^{***}$	-0.007	$(-4.72)^{***}$	-0.007	$(-4.31)^{***}$	-0.007	$(-5.23)^{***}$
SECURED		0.011	$(1.85)^{*}$	0.016	$(2.56)^{*}$	0.014	$(2.12)^{**}$	0.013	$(2.37)^{**}$
REVOLVER		-0.026	$(-3.30)^{***}$	-0.029	$(-3.58)^{***}$	-0.026	$(-2.96)^{***}$	-0.032	$(-4.59)^{***}$
GEN_COVENANTS		-0.004	(-0.77)	-0.013	$(-2.28)^{*}$	-0.011	$(-1.86)^{*}$	-0.004	(-0.86)
FIN_COVENANTS		-0.012	$(-2.28)^{**}$	-0.012	$(-2.14)^{*}$	-0.014	$(-2.38)^{**}$	-0.013	$(-2.41)^{**}$
PP_INDICATOR		-0.013	$(-2.10)^{**}$	-0.007	(-1.04)	-0.011	(-1.48)	-0.010	$(-1.81)^{*}$
LOG_PREVIOUS		0.004	(1.14)	0.006	(1.48)	0.002	(0.49)	0.004	(1.30)
PROFITABILITY		-0.014	$(-3.80)^{***}$	-0.130	$(-3.30)^{***}$	-0.158	$(-3.81)^{***}$	-0.014	$(-4.35)^{***}$
INTEREST_COVERAGE		0.000	(0.46)	0.000	(0.40)	0.000	(0.85)	0.000	(1.12)
FIRM_SIZE		-0.014	$(-4.59)^{***}$	-0.011	$(-3.27)^{***}$	-0.016	$(-4.13)^{***}$	-0.011	$(-3.87)^{***}$
LEVERAGE		-0.024	(-1.64)	-0.021	(-1.43)	-0.028	$(-1.82)^{*}$	-0.029	$(-2.14)^{**}$
Year fixed effects			Yes		Yes		Yes		Yes
N		7	4,140		3,749		4,140	4	,140
Adi. R^2			0.560		0.522		0.574	0	531

Determinants of Debt Monitoring Structure—Sensitivity Analysis

TABLE 4

Dependent variable is LEAD OWN (unless noted otherwise), the fraction of the deal owned by the lead arranger. All variables are defined in the Data Appendix. Each column number corresponds to the following analysis:

predict credit rating downgrades. In addition to change in quarterly earnings over the prior four quarters, the extended model regressor include changes in sales, number of industry (1) Extended DCV variable is alternatively measured as the Somers' D association statistic obtained from extended industry-specific (two-digit SIC codes) Probit regressions that segments reported, largest segment sales scaled by total sales, cash scaled by debt, and debt scaled by market capitalization. All variables are seasonally adjusted.

(2) Three Digit SIC Code DCV. DCV variable is measured at the three-digit SIC code level.
 (3) Timeliness/TLR at Firm Level. Regression is estimated using Timeliness and TLR measures computed at the firm level over the period 1985-2004.
 (4) Dependent Variable: LENDER-HERF. Dependent variable is LENDER HERF, the sum of the squared percentage ownership of each lender in the deal syndicate.

Significance levels of coefficient estimates are based on standard errors adjusted for clustering at the firm level. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively. based on a two-tailed test.

271

DEBT-CONTRACTING VALUE

Panel B: Exclusion of)	oan and deal	control vari	ables, lead arrang	ger and indus	try clustered sta D	mdard errors ependent Va	s, and industry e riable: <i>LEAD_O</i>	control variabl	es.		
	Predicted	Industr	ry Clustering	With Indı	ustry Controls	Industry R	andom Effects	Lead Arrang	er Fixed Effects	Sole Lei	nder Tobit
	Sign		1		2		3		4		5
DCV	I	-0.086	$(-3.17)^{***}$	-0.081	$(-3.78)^{***}$	-0.086	$(-4.07)^{***}$	-0.087	$(-4.11)^{***}$	-0.226	$(-5.89)^{***}$
TIMELINESS	I	-0.003	(-0.83)	-0.004	(-1.34)	-0.004	(-1.19)	-0.003	(-0.89)	-0.033	$(-4.28)^{***}$
TLR	I	-0.056	$(-2.12)^{**}$	-0.083	$(-4.44)^{***}$	-0.063	$(-3.80)^{***}$	-0.050	$(-3.06)^{***}$	-0.275	$(-8.65)^{***}$
UNRATED		0.027	$(3.30)^{***}$	0.027	$(4.28)^{***}$	0.027	$(4.25)^{***}$	0.022	$(3.47)^{***}$	0.017	(1.48)
NO_LEAD_PRIOR		0.025	$(4.38)^{***}$	0.025	$(4.83)^{***}$	0.025	$(4.80)^{***}$	0.026	$(5.06)^{***}$	0.052	$(6.30)^{***}$
LEAD_REPUTATION		-0.083	$(-13.92)^{***}$	-0.083	$(-13.77)^{***}$	-0.083	$(-13.77)^{***}$	-0.083	$(-13.53)^{***}$	-0.105	$(-9.71)^{***}$
DEAL_SIZE		-0.056	$(-11.31)^{***}$	-0.057	$(-15.30)^{***}$	-0.056	$(-15.06)^{***}$	-0.056	$(-14.92)^{***}$	-0.162	$(-24.78)^{***}$
LOAN_SPREAD		-0.004	(-0.71)	-0.004	(-0.77)	-0.004	(-0.79)	-0.004	(-0.89)	0.006	(0.75)
LOAN_MATURITY		-0.006	$(-4.77)^{***}$	-0.007	$(-4.57)^{***}$	-0.006	$(-4.54)^{***}$	-0.006	$(-4.3)^{***}$	-0.013	$(-5.32)^{***}$
SECURED		0.014	$(2.41)^{**}$	0.013	$(2.17)^{**}$	0.014	$(2.30)^{**}$	0.012	$(2.09)^{*}$	0.034	$(3.67)^{***}$
REVOLVER		-0.029	$(-3.34)^{***}$	-0.028	$(-3.59)^{***}$	-0.029	$(-3.68)^{***}$	-0.028	$(-3.72)^{***}$	-0.057	$(-4.75)^{***}$
GEN_COVENANTS		-0.009	(-1.59)	-0.009	$(-1.64)^{*}$	-0.009	$(-1.69)^{*}$	-0.015	$(-2.83)^{***}$	-0.104	$(-12.20)^{***}$
FIN_COVENANTS		-0.011	$(-2.23)^{**}$	-0.012	$(-2.25)^{**}$	-0.011	$(-2.07)^{**}$	-0.011	$(-2.03)^{*}$	-0.002	(-0.22)
PP_INDICATOR		-0.013	$(-2.03)^{**}$	-0.013	$(-1.98)^{**}$	-0.013	$(-1.96)^{**}$	-0.008	(-1.35)	-0.004	(-0.38)
LOG_PREVIOUS		0.005	(1.42)	0.004	(1.22)	0.005	(1.55)	0.004	(1.27)	0.010	$(1.71)^{*}$
PROFITABILITY		-0.149	$(-3.95)^{***}$	-0.144	$(-3.80)^{***}$	-0.146	$(-3.86)^{***}$	-0.129	$(-3.51)^{***}$	-0.415	$(-9.34)^{***}$
INTEREST_COVERAGE		0.003	(0.29)	0.000	(0.13)	0.000	(0.26)	0.000	(0.75)	0.001	$(2.38)^{**}$
FIRM_SIZE		-0.015	$(-3.24)^{***}$	-0.013	$(-4.33)^{***}$	-0.015	$(-4.85)^{***}$	-0.014	$(-4.53)^{***}$	0.007	(1.21)
LEVERAGE		-0.031	$(-2.69)^{***}$	-0.026	$(-1.76)^{*}$	-0.032	$(-2.19)^{**}$	-0.025	$(-1.79)^{*}$	-0.078	$(-3.29)^{***}$
Year fixed effects			Yes		Yes		Yes		Yes		les
Ν			4,140	4	l,140	7	ł,140		4,140	, С	206
Adj. R^2			0.567	0	.572	U	0.570		0.568	0.	771
Dependent variable i analysis:	s LEAD_OWN,	the fraction	of the deal owned	l by the lead a	urranger. All varia	ables are defir	ied in the Data /	Appendix. Each	ı column number e	corresponds to	the following
(1) Industry Clusterin,	g. Standard en	rors are cluste	ered at the two-dig	it SIC industr	y level.					:	

TABLE 4-Continued

(2) With Industry Controls. The regression is estimated after controlling for industry (two-digit SIC codes) averages of book-to-market, equity volatility and cash flow volatility. Coefficients of these industry-specific variables are not reported.

(3) Industry Random Effects. The regression is estimated with industry random effects (two-digit SIC codes) using GLS.
 (4) Lead Arranger Fixed Effects. The regression is estimated with lead arranger fixed effects.
 (5) Sole Lender Tobit. Tobit regression analysis where sample includes additional sole lender loans (*IEAD OWN* = 1).

Unless noted otherwise, significance levels of coefficient estimates are based on standard errors adjusted for clustering at the firm level *** **, and * indicate significance at the 1%, 5%, and 10% levels, respectively, based on a two-tailed test.

272 R. BALL, R. M. BUSHMAN, AND F. P. VASVARI

panel A, column 1 are consistent with our main results. We note that correlation between the measure generated by the extended *DCV* model and the measure generated by the more simplified "earnings-only" model is positive and significant (unreported Pearson correlation of 0.45).

Second, we attempt to reduce potential measurement errors in our proxies by re-estimating *DCV* at the three-digit industry level after requiring the existence of rating downgrades and a minimum of 20 observations (178 industries remain).³⁰ We also compute *TIMELINESS* and *TLR* at the firm level after requiring a minimum of five years of quarterly data. Again, the results presented in table 4, panel A, columns 2 and 3 are similar to those presented in the main table.

Finally, we use an alternative measure of syndicate monitoring incentives. Specifically, we replace *LEAD_OWN* with *LENDER_HERF*, which is computed as the sum of squared percentage ownership of both lead arrangers and participants (i.e., an ownership Herfindahl Index). Descriptive statistics on this variable are presented in table 1. This index measures the concentration of the overall syndicate ownership and is intended to capture the demand for joint monitoring activities by multiple members of the syndicate. Column 4 illustrates that our results are robust to this specification, which is not surprising given that *LEAD_OWN* and *LENDER_HERF* are highly correlated (Pearson correlation of 0.789).

5.4.2. Potential Econometric Misspecifications: Industry-Level Estimates of DCV and Correlated Omitted Variables with Respect to the Lead Arrangers. There are several potential econometric issues associated with our use of a two-digit SIC industry-level measure of the DCV variable. First, since our DCV variable is constant within a two-digit industry, there is concern that our standard errors may be underestimated due to intraindustry correlations of the error terms. To address this issue, we estimate standard errors to account for clustering at the industry level. Significance levels, reported in panel B, column 1, for each coefficient are robust to this alternative clustering specification.³¹

 $^{^{30}}$ We also estimate at the four-digit industry level (after applying similar filters). We run the *DCV* probit models for 243 industries. Results are similar except that the significance of the *DCV* variable drops to 5%. In addition, we estimate *DCV* models using both upgrades and downgrades and the results are the same. We do not report these sensitivity tests but the results are available upon request.

 $^{^{31}}$ As Petersen [2007] demonstrates, understatement of errors in our setting is affected by the multiplicative interaction of two statistics: (1) the amount of within-industry variation relative to the total variation in the explanatory variable, and (2) the amount of within-industry variation relative to the total variation in the error term. While our *DCV* variable, by definition, exhibits zero within-industry variation, unreported descriptive statistics reveal that our dependent and control variables, and thus the error term, do exhibit significant within-industry variation with respect to overall variation. Given our industry-specific explanatory variable, in the limit as the within-industry variation of our dependent and control variables goes to zero, our standard errors would approach the standard errors obtained from an industry average regression with

Second, it is also possible that our debt-contracting value proxies, measured at the industry level, capture industry-specific attributes that are not related to the informativeness of accounting (e.g., highly correlated with unspecified industry-specific risk factors or investment opportunity sets). To address this, we follow three approaches: (1) we estimate the main model using one-digit SIC industry fixed effects (unreported),³² (2) we add controls for within-industry averages of the book-to-market ratio, equity volatility, and cash flow volatility (table 4, panel B, column 2), and (3) we estimate the regression using generalized least squares with two-digit SIC industry random effects (table 4, panel B, column 3). All results are robust and our inferences are not changed.

Third, we estimate the regressions using lead arranger fixed effects.³³ The coefficients on the variables of interest and their significance do not change our inferences (see table 4, panel B, column 4). In addition, the significance levels do not change when we estimate standard errors clustered at the lead arranger level (we do not report these standard errors). Intra–lead arranger correlations of error term correlation might also be expected if only a few lead arrangers dominate in our sample (which is not the case).

It is also possible that lead arrangers use unobservable risk management techniques including credit derivatives (CDS) and securitization through collateralized loan obligations (CLO). While use of these mechanisms is becoming more prevalent, they were not very important during our sample period, 1992–2004. The trading of CDS on loans started in 2004, while standard documentation for the U.S. market was published by the International Swaps and Derivatives Association in June 2006 (Duncan [2006]). The first CLO completed by a U.S. bank occurred in late 1997. Total CLO volume for 1997–2001 (U.S. market) is estimated at around \$100 billion, less than 2% of the total amount of syndicated loans (see discussion in Ivashina [2007]).

Finally, we add sole lender loans, where the lending bank retains 100% of the loan, into our sample and estimate a tobit regression (we continue to adjust the standard errors for clustering at the firm level). Panel B, column 5 shows that our inferences from this expanded sample are robust.

only one observation per industry. However, because of the significant within-industry variation in the dependent and control variables with respect to their overall variation, we do not come close to this situation, and hence the robustness of our results.

³² Since our debt-contracting value of accounting information measures are defined at the two-digit industry level, we are unable to use two-digit industries fixed effects due to perfect collinearity.

³³ We perform two additional analyses that we do not report. First, we estimate the regressions only for the largest banks in the sample (approximately 63% of our deal observations). Second, we introduce a control variable computed as the ratio of the deal size to the total assets of the lead arranging bank (we manually collect total asset information from bank regulatory disclosures available on the WRDS system). Our inferences are not affected.

6. Empirical Analysis of Performance Pricing Provisions

6.1 EMPIRICAL ANALYSIS

Our second analysis investigates the choice of performance measure included in performance pricing provisions. We estimate the following probit model:

$$P(Ratio_PP = 1)$$

= $f\left(\alpha_0 + \alpha_1 DCV + \alpha_2 TIMELINESS + \alpha_3 TLR + \sum_{j=3}^{J} \beta_j Controls\right),$ (6)

where DCV, TIMELINESS, and TLR are the debt-contracting value of accounting information measures described in section 4. We include the same borrower-specific and loan-specific control variables used in the syndicate ownership analysis. Since equation (6) is estimated at the loan level, as opposed to the deal level, we include a number of additional loan-specific control variables. LOAN_SIZE is the logarithm of the dollar amount of the loan facility. RATING_GROUP is the loan-specific rating or borrower rating if the loan rating is missing. We transform the letter group ratings into numbers such that investment grade loan ratings (AAA to BBB ratings) are assigned smaller values from 1 to 4 while speculative grade loan ratings (BB to C ratings) are assigned larger values from 5 to 9. Finally, INCREASING_PP is an indicator variable equal to one if the loan contract contains an interestincreasing performance pricing provision to control for moral hazard costs in the choice of performance measures (see Asquith, Beatty, and Weber [2005]) and equal to zero otherwise. We expect that a contract is more likely to include an accounting-based performance measure when the interest rates are increasing.³⁴

6.2 SAMPLE SELECTION AND DESCRIPTIVE STATISTICS

Similar to the syndicate ownership sample, we apply several filters to obtain our final performance pricing loan sample (see table 5, panel A). Specifically, we match Dealscan data with Compustat, exclude financial firms, remove observations without control variables, remove all loans without performance pricing features, and require data availability on credit ratings to eliminate the possibility that performance pricing is based on accounting

³⁴ If the loan spread at the contract date is equal to the lowest spread in the pricing grid specified in the performance pricing provision, then the loan is considered to have an interest-increasing performance pricing provision. Similarly, if the loan spread at the contract date is equal to the highest spread in the pricing grid then the loan is considered to have an interest-decreasing performance pricing provision. If the loan spread at the contract date is neither equal to the highest spread nor equal to the lowest spread in the pricing grid then the loan is considered to have both interest-decreasing and interest-increasing performance pricing provisions (see Asquith, Beatty, and Weber [2005]).

	Simisius		
Panel A: Description of performance pricing sample selection pro	cedure.		
Performance Pricing Sample Selection (Loan Level)	Loans	Deals	Firms
Performance Pricing Sample (Loan Level)			
- Sample of syndicated loan contracts matched to Compustat	33,375	24,441	6,243
 Sample after excluding financial firms 	29,282	21,061	5,468
- Sample after requiring availability of loan and	17,819	12,483	4,002
borrower-specific variables			
 Sample with performance pricing provisions 	7,194	5,044	2,273
- Performance pricing sample after requiring ratings data in	4,203	2,928	1,232
Dealscan or Moody's Databases			
• Percentage with ratio-based performance pricing contracts	$54.3 \ \%$		
Percentage with rating-based performance pricing contracts	45.7~%		

TABLE 5

Performance Pricing Sample Selection and Descriptive Statistics

Panel B: Accounting based and ratio based measures used in performance pricing provisions.

		% in Total Number
Measure Description	Frequency	of PP Provisions
Accounting-based performance pricing provisions		
 Debt service coverage 	11	0.3~%
- Debt to tangible net worth	44	1.0~%
 Fixed charge coverage 	91	2.1~%
– Interest coverage	108	$2.5 \ \%$
– Leverage	142	3.3~%
- Senior debt to EBITDA	88	2.0~%
 Senior leverage 	7	0.2~%
- Total debt to EBITDA	1,742	40.5~%
- Tailored ratios (user conditions)	140	3.3~%
Ratings-based performance pricing provisions		
 Senior debt rating 	1,907	44.4~%
 Commercial paper rating 	18	0.4~%
Total number of performance pricing provisions	4,298	100.0~%
- Number of loans with two rating-based contracts	10	
 Number of loans with two accounting ratio–based contracts 	85	
Total number of loans in performance pricing sample	4,203	

(Continued)

ratios simply because rating agencies do not rate the firm or the loan.³⁵ The final sample comprises 4,203 (2,928) loans (deals) from 1,232 firms. Approximately 55% of the loans have performance pricing provisions based on accounting measures.

Table 5, panel B presents descriptive statistics on the measures used in performance pricing provisions. In this sample, the most commonly used accounting measure is debt to earnings before interest, taxes, depreciation, and amortization (EBITDA), which is consistent with Asquith, Beatty, and

 $^{^{35}}$ If loan ratings are missing, we use Moody's Historical Ratings Database to retrieve the issuer rating at the time of the loan. Issuer ratings are generally very close to loan ratings given that loans are the most senior form of debt.

Panel C: Sample distribution of model variables for performance pricing sample						
Ĩ	N (loans)	Mean	25%	Median	75%	
Dependent variable						
PP_RATIO	4,203	0.54	-	-	-	
Information variables						
DCV	4,203	0.347	0.253	0.333	0.407	
TIMELINESS (%)	4,203	1.196	0.483	0.937	1.747	
TLR	4,203	0.365	0.262	0.332	0.476	
Loan-specific variables						
LOAN_SIZE (\$ million)	4,203	447	100	230	500	
DEAL_SIZE (\$ million)	4,203	776	200	400	850	
LOAN_SPREAD (bps)	4,203	155	63	140	238	
LOAN_MATURITY	4,203	4.01	2.84	4.95	5.00	
SECURED	4,203	0.50	_	_	-	
RATING_GROUP	4,203	4.73	4	5	6	
REVOLVER	4,203	0.76	_	_	-	
LEAD_REPUTATION	4,203	0.81	-	-	-	
INCREASING_PP	4,203	0.48	-	-	-	
GEN_COVENANTS	4,203	4.39	2	4	7	
FIN_COVENANTS	4,203	2.42	1	2	3	
Borrower-specific variables						
PROFITABILITY	4,203	0.03	0.01	0.04	0.06	
INTEREST_COVERAGE	4,203	5.28	1.24	2.42	4.53	
FIRM_SIZE (\$ million)	4,203	4,868	637	1,559	4,014	
LEVERAGE	4,203	0.38	0.25	0.36	0.49	

TABLE 5—Continued

PP_RATIO is an indicator variable equal to 1 (equal to 0) if an accounting ratio (credit rating) based performance pricing provision is used in the syndicated loan contract. DCV is a credit market based earnings quality measure computed as the Somers' D association statistic obtained from industry specific (two-digit SIC codes) Probit regressions that predict credit rating downgrades. The downgrade predictors are the seasonally adjusted quarterly earnings over the prior four quarters. TIMELINESS is an equity market based earnings quality measure computed as the R^2 obtained from industry specific (two-digit SIC codes) pooled regressions of market-adjusted returns on quarterly earnings levels and seasonally differenced quarterly earnings. TLR is timely-loss recognition measured as the coefficient on negative returns in earnings-returns regressions estimated using quarterly data at the industry level (two-digit SIC codes). LOAN_SIZE is the individual loan size (\$ mil.). DEAL_SIZE is the size of the deal (\$ mil.). LOAN_SPREAD is the loan spread (in basis points). LOAN_MATURITY is the number of years to loan maturity. SECURED is an indicator variable equal to 1 if the loan is collateralized and 0 otherwise. RATING_GROUP is the loan rating (issuer rating if the loan rating is missing) coded from 1 for AAA/Aaa rated loans to 9 for C/C rated loans. REVOLVER is an indicator variable equal to 1 if the loan is a revolving loan and 0 otherwise. LEAD_REPUTATION is an indicator variable equal to 1 if the lead arranger is classified in the top 25 arrangers of syndicated loans in the U.S. in the year when the deal is signed. INCREASING_PP is equal to 1 if the loan contract has an interest increasing performance pricing provision and 0 otherwise. GEN_COVENANTS (FIN_COVENANTS) is the number of general (financial) covenants in the contract as reported by Dealscan. PROFITABILITY is the borrower's operating income before depreciation scaled by average total assets. INTEREST COVERAGE is the borrower's interest coverage ratio defined as the sum of interest expense and income before extraordinary items scaled by interest expense. FIRM_SIZE is the borrower's book value of total assets. LEVERAGE is the borrower's book value of debt (sum of debt in current liabilities and total long-term debt) divided by book value of total assets.

Weber [2005]. This ratio is present in 40.5% of the performance pricing provisions selected in our sample. Other accounting measures in our sample include leverage (3.3%), fixed charge coverage (2.1%), senior debt to EBITDA (2.0%), interest coverage (2.5%), debt to tangible net worth (1.0%), debt service coverage (0.3%), and senior leverage (0.2%). A small proportion of the performance pricing contracts (3.3%) are based on

tailored accounting measures (e.g., guarantees, contingent liabilities, and/or off balance sheet leases are included in the computation of debt levels), which are negotiated by the bank syndicate.

In our sample, we identify only two types of ratings in performance pricing contracts: senior debt ratings (44.4% in total number of provisions) and commercial paper ratings (only 0.4% in total number of provisions). The heavy use of senior debt ratings is consistent with the fact that most of the syndicated loans are negotiated as the most senior form of debt. Only a very small percentage of loans have performance pricing provisions based on more than one accounting measure or ratings type. Consistent with Asquith, Beatty, and Weber [2005], we find that interest-increasing contracts are more likely to be based on senior debt ratings than on debt-to-EBITDA or any other accounting ratio (unreported).

Table 5, panel C presents summary statistics for the performance pricing sample. We find no significant differences between the performance pricing sample and the syndicate ownership sample (see table 1, panel B) in terms of the debt-contracting value of accounting information proxies or firm-specific control variables. However, we note that the deal size is larger on average (\$776 million as opposed to \$458 million in the syndicated ownership sample). Also, loans in the performance pricing sample have more covenants (on average they have about four general covenants and two financial covenants). This suggests that it is important to control for the presence of covenants in multivariate tests given their inherent monitoring role that seems to complement performance pricing provisions.

We delete 122 loans (from 58 firms) from our sample that have both accounting- and rating-based performance pricing contracts. We manually investigate each of these loans by examining the financial statements of each borrower and, in most cases, find that the performance pricing is primarily based on a rating. However, these contracts specify that when ratings are not available, the performance pricing is based on an accounting measure.³⁶ We also delete loans with performance pricing based on other conditions, such as interest rates based on the age of the loan or on the percentage of the loan amount drawn down, unless the contract specifies an additional pricing grid based on either an accounting measure or a rating.³⁷

³⁶ For example, Pentair Inc.'s 10-K discloses the following: "On Sept. 2, 1999, the Company entered into two new revolving credit facilities aggregating \$800 million; a new five-year \$425 million revolving credit facility and a new 364-day \$375 million revolving credit facility. Inclusive of related facilities fees, the New Revolving Credit Facilities accrue interest at a floating rate based upon the rating of the Company's long term senior unsecured debt assigned by S&P and Moody's, or if no rating is available, based on a leverage ratio" (Note 6, Notes to Consolidated Financial Statements, filed with United States Securities Exchange Commission for the fiscal year ended December 31, 1999).

 $^{^{37}}$ The proportion with provisions based on these other conditions is less than 1% of the total number of contracts.

Probit Regression Coefficient Estimates (1-stat) for Predicting the Probability of Requiring Accounting-based Performance Pricing Provisions in the Syndicated Loan Contract TABLE 6

	Predicted				Dependent Vari	able: <i>PP_RATI</i>	0		
	Sign		1		2		3		4
DCV	+	0.199	$(3.56)^{***}$			0.192	$(3.07)^{***}$	0.177	$(2.76)^{***}$
TIMELINESS	+			0.091	$(1.58)^{*}$	0.032	(0.50)	0.089	(1.20)
TLR	+							0.100	$(2.70)^{***}$
LOAN SIZE		-0.241	$(-3.75)^{***}$	-0.218	$(-3.45)^{***}$	-0.246	$(-3.88)^{***}$	-0.237	$(-3.73)^{***}$
DEAL_SIZE		0.215	$(2.64)^{***}$	0.187	$(2.43)^{**}$	0.218	$(2.71)^{***}$	0.214	$(2.69)^{***}$
LOAN_SPREAD		0.561	$(4.99)^{***}$	0.540	$(4.87)^{***}$	0.561	$(4.99)^{***}$	0.579	$(5.10)^{***}$
LOAN_MATURITY		0.172	$(7.30)^{***}$	0.159	$(6.38)^{***}$	0.173	$(7.28)^{***}$	0.168	$(7.11)^{***}$
SECURED		0.573	$(5.23)^{***}$	0.561	$(5.52)^{***}$	0.565	$(5.31)^{***}$	0.548	$(5.17)^{***}$
RATING_GROUP		0.401	$(5.48)^{***}$	0.415	$(5.58)^{***}$	0.404	$(5.49)^{***}$	0.381	$(5.18)^{***}$
REVOLVER		0.246	$(2.85)^{***}$	0.227	$(2.55)^{**}$	0.247	$(2.84)^{***}$	0.242	$(2.74)^{***}$
LEAD_REPUTATION		-0.076	(-0.63)	-0.045	(-0.38)	-0.073	(-0.62)	-0.090	(-0.77)
INCREASING_PP		0.403	$(4.43)^{***}$	0.390	$(4.38)^{***}$	0.404	$(4.42)^{***}$	0.381	$(4.23)^{***}$
GEN_COVENANTS		-0.087	(-0.71)	-0.092	(-0.80)	-0.087	(-0.71)	-0.080	(-0.65)
FIN_COVENANTS		0.192	$(1.79)^{*}$	0.179	$(1.70)^{*}$	0.196	$(1.83)^{*}$	0.206	$(1.91)^{*}$
									(Continued)

DEBT-CONTRACTING VALUE

279

				COMMENSA				
Predicted				Dependent Vari	able: PP_RATI	0		
Sign		1		2		3		4
PROFITABILITY	-0.328	(-0.41)	-0.377	(-0.47)	-0.353	(-0.44)	-0.338	(-0.41)
INTEREST_COVERAGE FIRM_SIZE	0.929 - 0.467	$(2.10)^{**}$ $(-6.36)^{***}$	0.821 - 0.457	$(2.23)^{**}$ $(-6.55)^{***}$	0.924 - 0.466	$(2.16)^{**}$ $(-6.32)^{***}$	-0.468	$(1.86)^{*}$ $(-6.41)^{***}$
LEVERAGE	-0.458	$(-1.66)^{*}$	-0.505	$(-1.84)^{*}$	-0.452	$(-1.63)^{*}$	-0.425	(-1.50)
Year fixed effects		Yes		Yes		Yes		íes
Ν		4,203		4,203	4	1,203	4,	203
Likelihood ratio	35,	49.24^{***}	34	98.59***	355	0.41^{***}	3570	.69***
R^2		0.570		0.565	0	.570	0.	573
Somers' D		0.914		0.910	C	.914	0.	915
Dependent variable is <i>PP_RATIO</i> , equal to a credit market based earnings quality meas credit rating downgrades. The downgrade 1 quality measure computed as the R^2 obtaine differenced quarterly earnings. <i>TLR</i> is timel industry level (two-digit SIC codes). <i>LOANX</i> , (in basis points). <i>LOANMATURITY</i> is the m	o 1 (equal to 0 ure computed rredictors are ed from indust ly-loss recogni <i>IZE</i> is the logat umber of years	I) if an accounting rat as the Somers' D ass the seasonally adjust rry specific (two-digit tion measured as the tion measured as the rithm of the individual s to loan maturity, SC	tio (credit rati sociation statis ted quarterly c SIC codes) pc c coefficient or al loan size. <i>Dl</i>	ng) based performa tic obtained from in arnings over the pr obled regressions of n negative returns ii <i>EAL SIZE</i> is the logal rdicator variable equ	nce pricing prov dustry specific (ior four quarter market-adjusted 1 earnings-return ithm of the deal al to 1 if the loa	ision is used in the two-digit SIC codes s. <i>TIMELINES</i> is a Leturns on quarter ins regressions estim size. <i>LOAN</i> _SPREA, n is collateralized a	syndicated loan) Probit regression in equity marker the earnings levely ated using quar the logarith of 0 otherwise	contract. <i>DCV</i> is ions that predict t based earnings is and seasonally terly data at the m of loan spread <i>RATING. GROUP</i>

- Continued TARLF 6.

in the year when the deal is signed. INCREASING-PP is equal to 1 if the loan contract has an interest increasing performance pricing provision and 0 otherwise. GEN COVENANTS is the loan rating (issuer rating if the loan rating is missing) coded from 1 for AAA/Aaa rated loans to 9 for C/C rated loans. REVOLVER is an indicator variable equal to 1 if the loan (FIV.COVENANTS) is the number of general (financial) covenants in the contract as reported by Dealscan. PROFITABILITY is the borrower's operating income before depreciation scaled by average total assets. INTEREST. COVERAGE is the borrower's interest coverage ratio defined as the sum of interest expense and income before extraordinary items scaled is a revolving loan and 0 otherwise. *LEAD_REPUTATION* is an indicator variable equal to 1 if the lead arranger is classified in the top 25 arrangers of syndicated loans in the U.S. by interest expense. FIRM_SIZE is the logarithm of the borrower's book value of total assets. IEVERACE is the borrower's book value of debt (sum of debt in current liabilities and total long-term debt) divided by book value of total assets. Significance levels of coefficient estimates are based on standard errors adjusted for clustering at the firm level. ***, *** and

* indicate significance at the 1%, 5%, and 10% levels, respectively, based on a two-tailed test.

280

6.3 RESULTS

Table 6, column 1 documents that the coefficient on *DCV* is positive and significant, which is consistent with our prediction that lenders rely more on accounting-based monitoring and less on ratings-based monitoring when the debt-contracting value of accounting information increases. In other words, the timeliness provided by the accounting information is more important than the informativeness provided by credit ratings. Similar to prior specifications, the significance of all probit coefficient estimates in table 6 is computed based on standard errors clustered at the firm level.

This result is robust to an extensive set of firm-specific and loan-specific control variables. We try to control for reasons beyond debt-contracting value that banks may contract on accounting ratios instead of credit ratings in their pricing schedules. For example, banks may tend to use accounting ratios when borrowers are riskier as they place a premium on timeliness in these cases.³⁸ In this regard, we find that increases in LOAN_SPREAD and RATING_GROUP (proxies for the riskiness of the borrower) are associated with the choice of an accounting-based performance measure, consistent with the timeliness of a performance measure becoming relatively more important when sudden drops in credit quality are likely. It is of course also possible that the informativeness of credit ratings also varies with the risk of the firm. For example, credit ratings may be less informative for risky firms given the heightened uncertainty surrounding default. However, the higher demand for timeliness implied by risk and lower informativeness of credit ratings for riskier firms implies the same coefficient sign on the risk variables, so we cannot distinguish these stories.

Column 2 examines the effect of *TIMELINESS* on the lenders' optimal choice of performance measures in the performance pricing provision. We find that *TIMELINESS* is positive and marginally significant. When both *DCV* and *TIMELINESS* are present in the regression (column 3), we find that the presence of *DCV* reduces *TIMELINESS* to an insignificant level. In column 4, we document that both *TLR* and *DCV* are significant and positive.³⁹

In unreported tests, we also examine the association between accounting conservatism and the choice of accounting-based performance pricing provisions. We estimate pooled Basu [1997] and Ball and Shivakumar [2005] regressions with an additional interaction variable, which is an indicator variable that identifies firms with performance pricing contracts that use accounting ratios (Beatty, Weber, and Yu [2006] follow a similar approach). This analysis allows us to reduce measurement errors associated with the

³⁸ We thanks Tim Ritchie (head of global loans at Barclays Capital) for this observation.

³⁹ We repeat the probit analysis in table 6, column 4 using alternative industry-specific conservatism measures (scaled nonoperating accruals and special items; timeliness estimated using the Ball and Shivakumar [2005] model). Our inferences are not affected.

estimation of the asymmetric timeliness measure, but does not allow us to control for other loan-specific variables that are associated with the choice of the performance measure. The results suggest that firms with performance pricing based on accounting measures are more conservative in the years leading up to the loan date than firms with provisions using credit ratings. The coefficients on the interaction between negative returns and the presence of accounting-based performance pricing (Basu [1997] model) and the interaction between negative cash flows and the presence of accounting-based performance pricing (Ball and Shivakumar [2005] model) are positive and significant.

7. Conclusions

We investigate the central role played by publicly available accounting information in the optimal design of syndicated loan deals. Lead arrangers, by virtue of their exclusive relationship with the borrower, may ex ante possess private information about the borrower not known to other syndicate members. Loan participants also rely on lead arrangers to perform due diligence on the borrower before the loan is made. Such due diligence efforts are largely unobservable to syndicate participants. Finally, there is also a need to monitor borrowers on an ongoing basis after a loan deal has closed. Such information asymmetries create a demand for lead arrangers to hold a proportion of the loan that is increasing in the extent of adverse selection and moral hazard problems. We hypothesize that, as the debt-contracting value of accounting increases, lead arrangers are required to hold a smaller proportion of the syndicated loan deal.

We conjecture that the debt-contracting value of general purpose financial statements can mitigate adverse selection and moral hazard through a general transparency channel that is distinct from the formal contracting channel. More public transparency relative to a borrower's credit quality can reduce information asymmetries within the syndicate and so mitigate potential adverse selection and moral hazard problems. To isolate the general transparency channel in our empirical analysis, we control for the direct use of accounting variables in the formal contract.

There are, of course, other mechanisms available to deal with the adverse selection and moral hazard issues. We explore three characteristics that could potentially substitute for the debt-contracting value of accounting in determining the optimal fraction of the loan retained by the lead arranger. We hypothesize that accounting information with a high debt-contracting value is relatively more important in reducing the proportion of the loan retained by the lead arranger's reputation is low, and when the lead arranger has not previously served as a lead arranger for the same borrower.

Our primary measure for the debt-contracting value of accounting is the goodness-of-fit of a model of credit rating downgrades as a function of lagged seasonally adjusted accounting earnings. We document that the proportion

of the loan retained by the lead arranger is a decreasing function of the debtcontracting value of accounting data. We also document that the negative relation between the proportion retained by the lead arranger and debtcontracting value is larger when the borrower is not rated, when the lead arranger's reputation is low, and when the lead arranger has not previously served as a lead arranger for the same borrower.

Finally, we exploit the existence of performance pricing provisions in syndicated loan contracts to investigate how the debt-contracting value of accounting influences the choice of performance measure used in these provisions. We predict and find that, for loans that include performance pricing provisions, the likelihood that the single performance measure used in the provision is an accounting ratio rather than a credit rating increases as the debt-contracting value of accounting information improves.

An important contribution of our analysis is to more directly connect the existence of unresolved information asymmetries with direct, intuitive measures of the debt-contracting value of accounting information. This allows us to provide textured evidence on the central role of accounting information in the design of loan syndicates' ownership structures. Second, our empirical design distinguishes a general transparency channel through which accounting information operates to mitigate adverse selection and moral hazard that is distinct from a formal contracting channel, such as financial covenants or performance pricing provisions. This channel allows finer assessments of credit quality changes than covenants or interest rate triggers and determines the contracting use of accounting-based performance measures. Third, we empirically document that the relation between the debt-contracting value of accounting and the proportion of the loan retained by the lead arranger is stronger conditional on key aspects of the economic environment that indicate larger agency problems. This last result brings to the light important substitute relations between accounting information and other important mechanisms capable of reducing debtcontracting costs associated with information asymmetries.

Variable	Description	Data Source
Dependent variables		
LEAD_OWN	The fraction of the deal owned by the lead	Dealscan
	arranger.	
LENDER_HERF	A syndicate ownership Herfindahl Index defined	Dealscan
	as the sum of the squared percentage	
	ownership of each lender in the deal syndicate.	
PP_RATIO	An indicator variable equal to 1 (equal to 0) if an	Dealscan
	accounting ratio (credit rating) based	
	performance pricing provision is used in the	
	syndicated loan contract.	
Information variables		
DCV	A credit market–based earnings quality measure	Moody's/
	computed as the Somers' D association statistic	Compustat
	obtained from industry-specific (two-digit SIC	
	codes) probit regressions that predict credit	
	rating downgrades. The downgrade predictors	
	are the seasonally adjusted quarterly earnings	
	over the prior four quarters.	
TIMELINESS	An equity market–based earnings quality measure	CRSP/
	computed as the R^2 obtained from	Compustat
	industry-specific (two-digit SIC codes) pooled	
	regressions of market-adjusted returns on	
	quarterly earnings levels and seasonally	
	differenced quarterly earnings.	CDCD /
TLR	The coefficient on negative returns in	CRSP/
	earnings-returns regressions estimated using	Compustat
	quarterly data at the industry level (two-digit	
	SIC codes).	
Loan-specific and deal	-specific control variables	
UNRATED	An indicator variable that takes the value 1 if the	Dealscan/
NO LEAD DRIOD	firm or the loan is not rated.	Moody's
NO_LEAD_PRIOR	An indicator variable equal to 1 if the current lead	Dealscan
	arranger was not a lead arranger for the same	
I FAD PEDITATION	An indicator variable equal to 1 if the load	SDC Platinum
LEAD_REPUTATION	An indicator variable equal to 1 if the lead	SDC Plaulium
	syndicated loans in the United States according	
	to historical league tables provided by	
	Thompson Financial	
LOAN SIZE	The logarithm of the total dollar value of each	Dealscan
	loan.	Deulseun
DEAL_SIZE	The logarithm of the total dollar value of each	Dealscan
	deal.	
LOAN_SPREAD	The logarithm of the total annual all-in-spread	Dealscan
	drawn (in basis points) paid for each dollar	
	drawn down under the loan commitment	
	(including fees and interest).	
LOAN_MATURITY	The number of years to loan maturity.	Dealscan
SECURED	An indicator variable equal to 1 if the loan is	Dealscan
	secured with collateral.	

APPENDIX

(Continued)

Variable	Description	Data Source
REVOLVER	An indicator variable equal to 1 if the loan is revolving.	Dealscan
GEN_COVENANTS	The number of general covenants contained in the loan agreement.	Dealscan
FIN_COVENANTS	The number of financial covenants contained in the loan agreement.	Dealscan
PP_INDICATOR	An indicator variable equal to 1 if the loan contains a performance pricing provision.	Dealscan
RATING_GROUP	The loan rating (issuer rating if the loan rating is missing) coded from 1 for AAA/Aaa rated loans to 9 for C/C rated loans.	Dealscan/ Moody's
INCREASING_PP	An indicator variable equal to 1 if the loan contains an interest-increasing performance pricing provision.	Dealscan
DECREASING_PP	An indicator variable equal to 1 if the loan contains an interest-decreasing performance pricing provision.	Dealscan
Borrower-specific control	lvariables	
LOG_PREVIOUS	Logarithm of one plus the number of previous syndicated loans taken by the borrower.	Dealscan
PROFITABILITY	The sum of the borrower's prior four quarters' income before extraordinary items (data 8) scaled by average total assets (data 44) at the time of the deal.	Compustat
INTEREST_COVERAGE	The borrower's interest expense (data 22) plus income before extraordinary items (data 8) scaled by interest expense (data 22).	Compustat
FIRM_SIZE	The logarithm of the borrower's book value of assets (data 44) at the time of the deal.	Compustat
LEVERAGE	The borrower's debt in current liabilities (data 45) plus total long-term debt (data 51) scaled by book value of total assets (data 44).	Compustat

APPENDIX— Continued

REFERENCES

- AGHION, P., AND P. BOLTON. "An Incomplete Contracts Approach to Financial Contracting." *Review of Economic Studies* 59 (1992): 473–94.
- AHMED, A.; B. BILLINGS; R. MORTON; AND M. HARRIS. "The Role of Accounting Conservatism in Mitigating Bondholder-Shareholder Conflict Over Dividend Policy and in Reducing Debt Cost." *The Accounting Review* 77 (2002): 867–90.
- ALTMAN, E., AND G. SABATO. "Modelling Credit Risk for SMEs: Evidence from the U.S. Market." Abacus 43 (2007): 332–57.
- ALTUNBAS, Y.; B. GADANECZ; AND A. KARA. "The Evolution of Syndicated Loan Markets." The Service Industries Journal 26 (2006): 689–707.
- ASQUITH, P.; A. BEATTY; AND J. WEBER. "Performance Pricing in Bank Debt Contracts." Journal of Accounting & Economics 40 (2005): 101–28.
- BALL, R. "Infrastructure Requirements for an Economically Efficient System of Public Financial Reporting and Disclosure." Brookings-Wharton Papers on Financial Services, 4th Annual Conference, 2001.
- BALL, R.; A. ROBIN; AND G. SADKA. "Is Financial Reporting Shaped by Equity Markets or by Debt Markets? An International Study of Timeliness and Conservatism." *Review of Accounting Studies* (2008): Forthcoming.

- BALL, R., AND L. SHIVAKUMAR. "Earnings Quality in UK Private Firms: Comparative Loss Recognition Timeliness." Journal of Accounting & Economics 39 (2005): 83–128.
- BANK FOR INTERNATIONAL SETTLEMENTS (BIS). "The New Basel Capital Accord." Basel Committee on Banking Supervision, Third Consultative Paper. Basel, Switzerland: BIS, April 2003. Web site, http://www.bis.org/bcbs/bcbsp3.htm, 2003.
- BASU, S. "The Conservatism Principle and the Asymmetric Timeliness of Earnings." Journal of Accounting & Economics 24 (1997): 3–37.
- BEATTY, A.; J. WEBER; AND J. YU. "Conservatism and Debt." Working paper, MIT, 2006.
- BHARATH, S.; J. SUNDER; AND S. SUNDER. "Accounting Quality and Debt Contracting." *The Accounting Review* (2008): Forthcoming.
- BLACK, B. "The Legal and Institutional Preconditions for Strong Stock Markets." UCLA Law Review 48 (2001): 781-855.
- BOOT, A., AND A. THAKOR. "Can Relationship Banking Survive Competition?" *Journal of Finance* 55 (2000): 679–713.
- BUSHMAN, R.; Q. CHEN; E. ENGEL; AND A. SMITH. "Financial Accounting Information, Organizational Complexity and Corporate Governance Systems." *Journal of Accounting & Economics* 37 (2004): 167–201.
- BUSHMAN, R., AND A. SMITH. "Financial Accounting Information and Corporate Governance." Journal of Accounting & Economics 32 (2001): 237–333.
- DENNIS, S., AND D. MULLINEAUX. "Syndicated Loans." *Journal of Financial Intermediation* 9 (2000): 404–26.
- DIAMOND, D. W. "Financial Intermediation and Delegated Monitoring." *Review of Economic Studies* 51 (1984): 393–414.
- DIAMOND, D. W. "Monitoring and Reputation: The Choice Between Bank Loans and Directly Placed Debt." *Journal of Political Economy* 99 (1991): 689–721.
- DICHEV, I., AND D. SKINNER. "Large-Sample Evidence on the Debt Covenant Hypothesis." *Journal of Accounting Research* 40 (2002): 1091–123.
- DOVLE, J. "Credit Risk Measurement and Pricing in Performance Pricing-Based Debt Contracts." Ph.D. dissertation, University of Michigan, 2003.
- DUNCAN, A. "Loan-Only Credit Default Swaps: The March to Liquidity." *Commercial Lending Review*, September (2006): 19–22.
- FRANCIS, J.; R. LAFOND; P. OLSSON; AND K. SCHIPPER. "Costs of Equity and Earnings Attributes." *The Accounting Review* 79 (2004): 967–1010.
- FRANCOIS, P., AND F. MISSONIER-PIERA. "The Agency Structure of Loan Syndicates." Financial Review 42 (2007): 227–45.
- GIVOLY, D., AND C. HAYN. "The Changing Time-Series Properties of Earnings, Cash Flows and Accruals: Has Financial Reporting Become More Conservative?" *Journal of Accounting and Economics*, 29 (2000): 287–320.
- GORTON, G., AND G. PENNACCHI. "Banks and Loan Sales: Marketing Nonmarketable Assets." Journal of Monetary Economics 35 (1995): 389–411.
- GUAY, W., AND R. VERRECCHIA. "Discussion of an Economic Framework for Conservative Accounting by Bushman and Piotroski (2006)." *Journal of Accounting & Economics* 42 (2007): 149–65.
- HOLMSTRÖM, B. "Moral Hazard and Observability." The Bell Journal of Economics 10 (1979): 74-91.
- HOLMSTRÖM, B. "Moral Hazard in Teams." The Bell Journal of Economics 13 (1982): 324-40.
- HOLMSTRÖM, B., AND J. TIROLE. "Financial Intermediation, Loanable Funds, and the Real Sector." *The Quarterly Journal of Economics* 112 (1997): 663–91.
- IVASHINA, V. "The Effects of Syndicate Structure on Loan Spreads." Working paper, Harvard Business School, 2007.
- JENSEN, M., AND W. MECKLING. "Theory of the Firm: Managerial Behaviour, Agency Costs and Ownership Structure." *Journal of Financial Economics* 3 (1976): 305–60.
- JONES, J.; W. LANG; AND P. NIGRO. "Agent Bank Behavior in Bank Loan Syndications." Journal of Financial Research 28 (2005): 385–402.

- LEE, S., AND D. MULLINEAUX. "Monitoring, Financial Distress, and the Structure of Commercial Lending Syndicates." *Financial Management* 33 (2004): 107–30.
- LEFTWICH, R. "Accounting Information in Private Markets: Evidence From Private Lending Agreements." *Accounting Review* 58 (1983): 23–42.
- LELAND, H., AND D. PYLE. "Informational Asymmetries, Financial Structure, and Financial Intermediation." *Journal of Finance* 32 (1977): 371–87.
- MOODY'S INVESTOR SERVICES. "Global Medical Products and Device Industry." Moody's Investor Services Report, Rating Methodology, August 2006. New York: Moody's Investor Services, 2006.
- PANYAGOMETH, K., AND G. ROBERTS. "Private Information, Agency Problems and Determinants of Loan Syndications: Evidence From 1987–1999." Working paper, York University, 2002.
- PETERSEN, M. "Estimating Standard Errors in Finance Panel Data Sets: Comparing Approaches." Working paper, Northwestern University, 2007.
- PICHLER, P., AND W. WILHELM. "A Theory of the Syndicate: Form Follows Function." Journal of Finance 56 (2001): 2237–64.
- SCHIPPER, K. "Fair Values in Financial Reporting." Web site, http://fars.org/ 2005AAAFairValue, 2005.
- SIMONS, K. "Why Do Banks Syndicate Loans?" New England Economic Review of the Federal Reserve Bank of Boston (1993): 45–52.
- SMITH, C., AND J. WARNER. "On Financial Contracting: An Analysis of Bond Covenants." Journal of Financial Economics 7 (1979): 117–61.
- SOMERS, R. "A New Asymmetric Measure of Association for Ordinal Variables." American Sociological Review 27 (1962): 799–811.
- SUFI, A. "Information Asymmetry and Financing Arrangements: Evidence From Syndicated Loans." *Journal of Finance* 62 (2007): 629–68.
- VASVARI, F. "Managerial Incentive Structures, Conservatism and the Pricing of Syndicated Loans." Working paper, London Business School, 2006.
- WATTS, R. "Conservatism in Accounting, Part I: Explanations and Implications." *Accounting Horizons* 17 (2003a): 207–21.
- WATTS, R. "Conservatism in Accounting, Part II: Evidence and Research Opportunities." Accounting Horizons 17 (2003b): 287–301.
- WATTS, R., AND J. ZIMMERMAN. *Positive Accounting Theory*. Englewood Cliffs, NJ: Prentice-Hall, Inc., 1986.
- WITTENBERG-MOERMAN, R. "The Impact of Information Asymmetry on Debt Pricing and Maturity." Working paper, The Wharton School, 2006a.
- WITTENBERG-MOERMAN, R. "The Role of Information Asymmetry and Financial Reporting Quality in Debt Contracting: Evidence From the Secondary Loan Market." Working paper, The Wharton School, 2006b.
- ZHANG, J. "Efficiency Gains From Accounting Conservatism: Benefits to Lenders and Borrowers." Working paper, MIT, 2004.